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Preface

Today, designing of machines and dies is done to a large extent with the help of computer
programs. However, the predictions of theses programs do not always agree with the
practical results, so that there is a need to improve the underlying mathematical models.
Therefore, knowledge of the formulas, on which the models are based and the limits of
their applicability is necessary if one wants to develop a new program or improve one
already in use.

Often the plastics engineer has to deal with different fields of engineering. The search for
the appropriate equations in the various fields concerned can be time-consuming.
A collection of formulas from the relevant fields and their applications, as given in this
book, make it easier to write one’s own program or to make changes in an existing program
to obtain a better fit with the experiments.

It is often the case that different equations are given in the literature on plastics technology
for one and the same target quantity. The practicing engineer is sometimes at a loss
judging the validity of the equations he encounters in the literature,

During his long years of activity as an R&D engineer in the polymer field at the BASF AG
and other companies, Natti Rao tested many formulas published while solving practical
problems. This book presents a summary of the important formulas and their appli-
cations, which Natti Rao, in cooperation with the well-known resin and machine
manufacturers, successfully applied to solve design and processing problems.

The formulas are classified according to the fields, rheology, thermodynamics, heat
transter, and part design. Each chapter covers the relevant relations with worked-out
examples. A separate chapter is devoted to the practical equations for designing extrusion
and injection molding equipment with detailed examples in metric units.

In addition, this work contains new, straightforward, practical relationships that have
been developed and tested in recent years in solving design problems in the area of
extrusion and injection molding.

The topic of polymer machine design has been dealt with in several books, However, in
these books the know-how was presented in a way that the vast majority of plastics
engineers cannot easily apply it to the problems in their day-to-day work. By means of
thoroughly worked-out, practical examples this book elucidates the computational
background of designing polymer machinery in a manner which every engineer can
understand and easily apply in daily practice.

We wish to express our thanks to our colleagues at the University of Massachusetts at
Lowell, USA, for fruitful discussions, Qur thanks are also due to Faculty Innovation Center
of the University of Austin, Texas, USA, for help in preparing the manuscript.

Austin, USA Natti 5. Rao, Ph. D.
Karlsruhe, Germany Giinter Schumacher, Ph. D.
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A Final Word

The aim of this book is to present the basic formulas of rheology, thermodynamics, heat
transfer, and strength of materials applicable to plastics engineering and to show how,
starting from these formulas, models for designing polymer processing equipment can
be developed.

Thoroughly worked out examples in metric units illustrate the use of these formulas,
which have been successfully applied by well known machine manufacturers time and
again in their design work. However, owing to the ever increasing growth of knowledge
brought forth by research and development in the plastics field, a book of this kind needs
to be renewed often and as such cannot claim to be an exhaustive work.



1 Formulas of Rheology

One of the most important steps in processing polymers is melting the resin, which is
initially in the solid state, and forcing the melt through a die of a given shape. During this
operation, the melt, whose structure plays a key role in determining the quality of the
product to be manufactured, undergoes different flow and deformation processes.

The plastics engineer has therefore to deal with the melt rtheology, which describes the flow
behavior and deformation of the melt. The theory of elasticity and hydromechanics can
be considered the frontier field of rheology, because the former describes the behavior of
ideal elastic solids, whereas the latter is concerned with the behavior of ideal viscous fluids.

Ideal elastic solids deform according to Hooke’s Law and ideal viscous fluids obey the
laws of Newtonian flow. The latter are also denoted as Newtonian fluids, Plastic melts
exhibit both viscous and elastic properties.

Thus, the design of machines and dies for polymer processing requires quantitative
description of the properties related to polymer melt flow. Starting from the relationships
for Hookean solids, formulas describing viscous shear flow of the melt are treated first,
as far as they are of practical use in designing polymer machinery. This is followed by a
summary of expressions for steady and time-dependent viscoelastic behavior of melts.

1.1 Ideal Solids

The behavior of a polymer subjected to shear or tension can be described by comparing
its reaction to external force with that of an ideal elastic solid under load. To characterize
ideal solids, first of all it is necessary to define certain quantities as follows [1]:

The axial force E, in Figure 1.1 causes an elongation Al of the sample of diameter d, and
length ; that is fixed at one end. Following equations apply for this case:

Engineering strain:

&=— (1.1)
L
Hencky strain:
l
£=In|— (L.2)
“[%]

Tensile stress:

o, = (1.3)

5
4
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Figure 1.1  Deformation of a Hookean solid by a tensile stress [1]

af

Reference area:

zd;
=—= 1.4
, (1.4)
Poisson’s ratio:
_ Adld,
Alll,

Figure 1.2 shows the influence of a shear force F,, acting on the area A of a rectangular
sample causing the displacement AU. The valid expressions are defined by:

(1.5)

Shear strain:

AU
y=— (1.6)
i
Shear stress:
E
7= It (1.7)

| A Y

/ ¥4

i /!
/.

Figure 1.2 Deformation of a Hookean solid by shearing stress [1]




1.2 Newtonian Fluids 3

Figure 1.3 Hookean solid under compression [1]

The isotropic compression due to the pressure acting on all sides of the cube shown in
Figure 1.3 is given by the engineering compression ratio &

AV
x=—o

v, (1.8)

where AV is the reduction of volume of a body with the original volume V, due to
deformation.

1.1.1 Hooke's Law

The linear relationships between stress and strain of a Hookean solid are given by [1].

o, =E-¢ (1.9)
T=G-y (1.10)
p=-K-x (1.11)

Where E is the modulus of elasticity, G is the shear modulus, and K is the bulk modulus.
These moduli are constant for a Hookean solid. In addition, the relationship existing
between E, G and K is expressed as {1]

E=2G{+m=3K(1-24) (1.12)
For an incompressible solid this leads (K — o, gt — 0.5) to {1]

E=3G (1.13)

1.2  Newtonian Fluids
There is a linear relationship between stress and strain in the case of Newtonian fluids
similar to the one for ideal elastic solids.

The fluid between the upper plate in Figure 1.4 is moving at a constant velocity U, and
the lower stationary plate experiences a shear stress 7 (see also Figure 1.2).
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L.

Figure 1.4  Shear flow

Lt

T= (1.14)
The shear or deformation rate of the fluid is equal to
y= & = d_u (1.15)
H dy
The shear viscosity is defined as
n== (1.16)
4
For an extensional flow, which corresponds to the tension test of a Hookean solid, we get
o, =A£ (1.17)
where
7 = normal stress
A = Trouton viscosity
£ = strain rate
Analogously to Equation 1.13 one obtains
A=3n (1.18)

1.3  Formulas for Viscous Shear Flow of Polymer Melts

Macromolecular fluids such as thermoplastic melts exhibit significant non-Newtonian
behavior. This can be seen in the marked decrease of melt viscosity when the melt is
subjected to shear or tension as shown in Figure 1.5. The plastic melt in the channels of
polymer processing machinery is subjected mainly to shear flow. Therefore, knowledge
of the laws of shear flow is necessary when designing machines and dies for polymer
processing. For practical applications, the following summary of the relationships was
found to be useful,
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g€, 195% —
Figure 1.5  Tensile viscosity and shear rate viscosity of a polymer melt as a function of strain rate [21]

1.3.1  Apparent Shear Rate
The apparent shear rate for a melt flowing through a capillary is defined as

.=4Q

xR

(1.19)
where Q is the volume flow rate per second and R is the radius of the capillary.

1.3.2 Entrance Loss

Another rheological quantity of practical importance is the entrance loss p,, representing
the loss of energy of flow at the entrance to a round nozzle. This is empirically correlated
by the relation [2]

po=c " (1.20)

Table 1.1  Resin-Dependent Constants ¢ and m in Equation 1.20 [2]

Polymer C m

Polypropylene (Novolen 1120 H) 2.551- 107 2.116
Polypropylene (Novolen 1120 L) 1.463 - 107 1.976
Polypropylene (Novolen 1320 L) 2.871 - 107 2.469
LDPE (Lupolen 1800 M) 1.176 - 107 1.434
LDPE (Lupolen 1800 S) 6.984 - 10° 1.072
LDPE (Lupolen 1810 D) 5.688 - 107 1.905
HDPE (Lupolen 6011 L} 3,940 107 1.399
HDPE (Lupolen 6041 D} 1.778 - 10’ 1.187
Polyisobutylene {Oppanol B 10) 6.401-10° 1.575

Polyisobutylene {Oppanol B 15) 1.021- 107 2614
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where ¢ and m are empirical constants and 7 is the shear stress. These constants can be
determined from the well-known Bagley-curves as shown in Figure 1.7. The values of
these constants are given in Table 1.1 for some thermoplastic materials. Shear stress and
entrance loss are measured in Pa in the calculation of ¢ and m.

1.3.3  True Shear Stress

The flow curves of a particular LDPE measured with a capillary rheometer are given in
Figure 1.6. The plot shows the apparent shear rate ¥, as a function of the true shear stress
7 at the capillary wall with the melt temperature as a parameter. The entrance loss p, was
obtained from the Bagley plot shown in Figure 1.7.

10*

F=230°C
!
180%C

'/rsn'c
10? A

10!
/

0
Vi 10* 0 gy 10°

Shegr slress T —e=

A

Shear rote y —=

Figure 1.6  Flow curves of a LDPE [8]

B0
* £=1mm bor
° £=0,6mm

Pressure g —e=

Capillary geometry (/R —e=
Figure 1.7 Bagley plots of a polystyrene with the capillary length L and radius & [3]
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Thus, the true shear stress 7is given by
r=f P (1.21)
2(L/R)
where

L =length of the capillary
R =radius of the capillary
p = pressure of the melt (see Figure 1.39).

1.3.4 Apparent Viscosity
The apparent viscosity 7, is defined as

T
’?a =— ( 122)
7a

and is shown in Figure 1.8 as a function of shear rate and temperature for a LDPE.

i
"‘*‘5'\ 12150°C
RN

} 10 <
= \
= Bt \
- \\\
"]‘ ] ] F 3 &
0 0 0 ¢ N

5
Sheor rate § —e=

Figure 1.8  Viscosity functions of a LDPE [8]

1.3.5 True Shear Rate

The true shear rate ¥, is obtained from the apparent shear rate by applying the correction
for the non Newtonian behavior of the melt according to Rabinowitsch

. _[(nt+3])
?’t—[ " )?’a {1.23)

The meaning of the power law exponent # is explained in the Section 1.3.7.2.
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10%
Pa:s T
=190, Polystyrene
=} o 20
z &
sE W NN
‘%E — Tl RN
=== (7} 3
10? ! :\*}
1" 100 0 ? 0

Apporent shear rote g, —=—
True sheor rate i —=

Figure 1.9 True and apparent viscosity functions of a polystyrene at different temperatures [2]

1.3.6  True Viscosity
The true viscosity 1, is given by

My = — (1.24)

Y

In Figure 1.9, the true and apparent viscosities are plotted as functions of the correspond-
ing shear rates at different temperatures for a polystyrene. As can be seen, the apparent
viscosity function is a good approximation for engineering calculations.

1.3.7  Empirical Formulas for Apparent Viscosity

Various fluid models have been developed to calculate the apparent shear viscosity 7,
[9}. The following sections deal with some important relationships frequently used in
design calculations.

1.3.7.1 Hyperbolic Function of Prandtl and Eyring

The relation between shear rate %, and shear stress 7 according to the fluid model of
EvriNG [4] and PranDTL [5] can be written as

¥, = Csinh(z/A) (1.25)
where € and A are temperature-dependent material constants.

The evaluation of the constants Cand A for the flow curve of LDPE at 190 °C in Figure 1.10
leads to C=4 5™ and A = 3 - 10 N/m?. It can be seen from Figure 1.10 that the hyperbolic
function of Eyring and Prandtl holds good at low shear rates,
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10
5“'
0
' 3
o~ '}:’
'E 10! J
& /
mi
PE-LD
190%C
‘]D
3 &
w0 0o W

Shear stress T —e
Figure 1.10  Comparison between measurements and values calculated with Equation 1.25 [8]

1.3.7.2 Power Law of Ostwald and de Waele

‘The power law of Ostwalp [6] and pE WaeLE [7] is easy to use, hence widely employed in
design work [10]. This relation can be expressed as

-, =K7" (1.26)
or
A=Kl (1.27)

where K denotes a factor of proportionality and n the power law exponent. Another
form of power law often used is

7=Kg 73R (1.28)
or
7=Ke| 7227 7o (1.29)

In this case, ny is the reciprocal of nand Ky = K™% . From Equation 1.26 the exponent
1 can be expressed as

dlgy,
=8% 1.30
" Qigr (1.30)



10 1 Formulas of Rheology [Refs. on p. 33]

0} |
gt PE-LD /
L
0
mﬂ /
n! [t{
/ om0

. /

/

Shear rate 7 —e

10—~

i

1 ] 3 : 5 8
gl L] L 10 U] P 1

Shear stress T —=
Figure 1.11  Determination of the power law exponent n in Equation 1,30

As shown in Figure 1.11, in a double log plot the exponent n represents the local gradient
of the curve ¥, vs. 7.

Furthermore
1_ dlgr _dign, +dlgy, _dlgm, (1.31)
n dlgy, dlg¥, dlg7,

The values of K and n determined from the flow curve of LDPE at 190 °C shown in
Figure 1.12 were found to be K = 1.06 - 107 and » = 2.57 [8]. As can be seen from
Figure 1.12, the power law fits the measured values much better than the hyperbolic
function of EYRING {4} and PRANDTL [5]. The deviation between the results from the power
law and from the experiment is a result of the assumption that the exponent # is constant
throughout the range of shear rates considered, whereas in fact » varies with the shear
rate. The power law can be extended to consider the effect of temperature on the viscosity
as follows:

7, = Kog - exp(-f3-T)- 27 (1.32)
where

Kog = consistency index
B = temperature coefficient
T = temperature of melt,
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10

Shear rale y —=
=

ﬁ‘
PE-L0
190°C

i B

0* w10

Shegr slress T —=
Figure 1.12 Comparison between measured values and power law

Example
Following values are given for a LDPE:
np = 03286
B =0.00863(°C™")
Kop =135990 (N -s™® -m™%)
The viscosity 7, at 7= 200°C and #, =500s" is calculated from Equation 1.32

1, =3731Pa-s

1.3.7.3 Muenstedt’s Polynomial
The fourth degree polynomial of MueNsTEDT [2] provides a good fit for the measured
values of viscosity. For a specific temperature this is expressed as

lgm, = Ag + A lg7, + A; (187, + A, (17,0 + A, (Ig7)" (1.33)

where Ay, A, A,, As, A, represent resin-dependent constants. These constants can be
determined with the help of the program of Rao [10], which is based on multiple linear

regressions.

This program in its general form fits an equation of the type
y=ayta s ta;x,+...ta,x,

and prints out the coefficients a,, a, and so on for the best fit.
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Shift Factor for Crystalline Polymers

The influence of temperature on viscosity can be taken into account by the shift factor a

{2].
For crystalline polymers this can be expressed as

ar = b(Ty) exp(b,/T) (1.34)

where

b,, b, = resin dependent constants

T = melt temperature (K)

T, = reference temperature (K)
Shift Factor for Amorphous Polymers

The shift factor 4, for amorphous polymers is derived from the WLF equation and can
be written as

(1.35)

where

> ¢; = resin dependent constants
T = melt temperature (°C)
Ty = reference temperature (°C)

The expression for calculating both the effect of temperature and shear rate on viscosity
follows from Equation 1.33.

lgn, =lgar + Ay + A, lglay 7,)+ A, Uglar 7,)F

o e (1.36)
+ Ay [lglar 7,01 + A, Ulglar 7)1
With Equation 1.31 we get
1 . . X
— =1+ A, +2 A4, lglar 7,)+ 3 A, [lg(ar 7,)F +4 A, [lg(ar 7,07 (1.37)

1

The power law exponent is often required in the design work as a function of shear rate
and temperature. Figure 1.13 illustrates this relationship for a specific LDPE. The curves
shown are computed with Equation 1.34 and Equation 1.37. As can be inferred from
Figure 1.13, the assumption of a constant value for the power law exponent holds good
for a wide range of shear rates.
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Figure 1.13  Power law exponent of a LDPE as a function of shear rate and temperature

Example
The viscosity for a LDPE is to be calculated with the following constants:

A = 4.254]
Al = =0.4978
A2 =-0.0731
A3= 00133
Ad=-0,0011
B = 513-10°
B2= 5640 K

at #,=500s" and T'= 200 °C.
Solution
ay from Equation 1.34
5640
=5.13-10"°  exp| =— | = 0.774
o exP( 473 ]
With
X =lglar - 7,)

X =1g(0.774 - 500) = 2.588
11, from Equation 1.36

n, =10(lgar + Ag + A X+ A, X2+ A, X2+ A, XY
Substituting the values of Ay, A, and so on yields

7}, =351.78Pa s
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The power law exponent is obtained from Equation 1.37
n=(+A+2A, X+34, X* +44, X3

Using the values for Ay, A}, and so on
n=3.196

1.3.7.4 Carreau’s Viscosity Equation [11]

As shown in Figure 1.14 {12}, the Carreau equation provides the best fit for the viscosity
function, reproducing the asymptotic form of the plot at high and low shear rates correctly.

The equation is expressed as
_ A
1+B-7,)°’

where A, B, C are resin-dependent constants. By introducing the shift factor 4y into
Equation 1.38, the temperature-invariant form of the Carreau equation can be given as

(1.38)

A

Aa
=71 1.39
T G+ Bay 7. 1.39)

For a number of resins the shift factor can be calculated as a function of temperature

from the following equation with good approximation [9, 10]
8.86(T, —Tyy)  8.86(T = Tyr)

1016 + (T, - Tyr) 1016 +(T; — Ty

where T, (°C) is the temperature at which the viscosity is given and T, (°C) the temperature
at which the viscosity is calculated.

(1.40)

Igar(T;,T,) =

The standard temperature Ty is given by {9)
Tgr=T,+50°C (1.41)

L

L Slope:-C

Viscosity 7 —e=

7=

Shear rate J —e

Figure 1.14  Determination of Carreau-parameters from a viscosity function [12]
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Data on typical glass transition temperatures of polymers are given in Table 3.1 [9].

The power law exponent n can be obtained from Equation 1,3%:
l=_c.B'—“T_.Jya+1 (1.42)
n l1+B-ap ¥,

For high shear rates n becomes [12]

1
n=

1-C
Example
Following constants are given for a specific LDPE:
A =32400Pa-s
B = 31s
C =062
TST = —133 DC

The viscosity is to be calculated at
T,=200°Cand ¥, = 5005

Solution

From Equation 1.40 one obtains

886(T - Tyr) __8861190-(-133)] _ .,
1016+ (T, —Ter) 1016 +[190 —(-133)]

and
__886(T -Ty) _ 886{200-(-133)] _ o
1016+ (T, - Tep)  101.6 +[200— (-133)]
ar=107=10"%=0.89
The power law exponent is calculated from Equation 1.42
~1 -1
. (—c-z H) - [—0.62 113795 1} 263
1+ Z 1+1379.5
where Z=B-ag -, =3.1-0.89 - 500 = 1379.5.
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1.3.7.5 Klein's Viscosity Formula [14]
The regression equation of Klein et al. [14] is given by

g, =ag+a,Inf, +a, Iy +a, T+a, T +a, T-lng, (1.43)

T = temperature of the melt (°F)
1), = viscosity (Ib; - s/in’)

The resin-dependent constants g, to a,, can be determined with the help of the computer
program given in [10], as in the case of the A-coefficients in Equation 1.33.
Example

Following constants are valid for a specific type of LDPE. What is the viscosity 1, at
¥, = 5005 and T=200°C?

a, = 3.388

a, =-6.351-10""

a;, =-1.815-107"

a, =-5975-10"

a,, =-2.51 -107° and
a, = 5.187-107
Solution

T{°F)=18-T(*CH32=1.8-200+ 32 =392
With the constants above and Equation 1.43 one gets
1, = 0.066 Ib; - s/in”
and in SI-units
1, = 6857 - 0.066 = 452.56 Pa - s

The expression for the power law exponent # can be derived from Equation 1.31 and
Equation 1.43. The exponent » is given by

1
Putting the constants a, ... a, into this equation obtains
n=2919

1.3.7.6 Effect of Pressure on Viscosity

Compared to the influence of temperature, the effect of pressure on viscosity is of minor
significance,
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Table 1.2  Effect of Pressure on Viscosity for Polystyrene, Equation 1.46

P 7,
bar

30 103 7,
100 1105 77,
200 1.221 7,
300 135 7,
500 165 7,
1000 272 7,
3000 20 7

However, the relative measure of viscosity can be obtained from [9, 15, 16]
7, =1, expla, - p) (1.45)
where

1, = viscosity at pressure p and constant shear stress 7,
Tlo = viscosity at constant shear stress 7
o, = pressure coefficient

For styrene polymers 7, is calculated from [14]

p
= —_— 1.46

where p = pressure in bar.

Thus the change of viscosity with pressure can be obtained from Equation 1.46. Table 1.2
shows the values of viscosity calculated according to Equation 1.46 for a polystyrene of
average molecular weight. It can be seen that a pressure of 200 bar causes an increase of
viscosity of 22% compared to the value at 1 bar. The pressure coefficient of LDPE is less
than that of PS by a factor of 3—4 and the value of HDPE is again smaller by a factor of 2
compared to LDPE. This means that in the case of polyethylene an increase of pressure
by 200 bar would enhance the viscosity only by 3 to 4%. Consequently, the effect of
pressure on viscosity can be neglected in the case of extrusion processes which generally
use low pressure. However, in injection molding with its high pressures, typically the
dependence of viscosity on pressure has to be considered.

1.3.7.7 Dependence of Viscosity on Molecular Weight

The relationship between viscosity and molecular weight can be described by [12]

n, = K' 533 (1.47)
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where
M,, = molecular weight

W

K =resin dependent constant

The approximate value of K for LDPE is
K=228-10"

and for polyamide 6
K=521-107"

according to the measurements of Laun [21].

These values are based on zero viscosity.

1.3.7.8 Viscosity of Two Component Mixtures

The viscosity of a mixture consisting of a component A and a component B can be
obtained from [17]

lgmy =Calgn, +Cylgng (1.48)
where

1 = viscosity
C = weight per cent

Indices:

M:  mixture
A, B: components

1.4  Viscoelastic Behavior of Polymers

Polymer machinery can be designed sufficiently accurately on the basis of the relationships
for viscous shear flow alone. However, a complete analysis of melt flow should include
both viscous and elastic effects, although the design of machines and dies is rather difficult
when considering melt elasticity and therefore rarely used. Waener [18] and FiscHER
[19] made attempts to dimension dies taking elastic effects into account.

For a more complete picture of melt theology the following expressions for the viscoelastic
quantities according to LauN [20, 21] are presented. The calculation of the deformation
of the bubble in film blowing is referred to as an example of the application of Maxwell’s
model.
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1.4.1 Shear
1.4.1.1 Linear Viscoelastic Behavior

The linear viscoelastic behavior occurs at low shear rates or small shear,

(1.49)

Steady Shear Flow
Zero viscosity 7], as a material function for the viscous behavior (Figure 1.15 and 1.16):
%
==
h
| TN | e
s / %
* I ! i} BH—
e
E K
Time } —=-

Figure 1.15 Time dependence of shear strain and shear stress in a stress test at constant shear rate
#, and subsequent recoil due to unloading (shear stress t=0) with y,; as recoverable

shear strain [21]
%
g
e
{ TN | s
~ - A
=3 1
2 |
w1 1
1 I ; i
] L
Time ¢ —=

Figure 1.16  Creep test at constant shear stress 7, and subsequent retardation after unloading

(shear stress t=0) [20].

};s = recoverable shear strain; | = initial state; I = steady state; lll = retardation
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Steady state shear compliance J{ (Figure 1.15 and Figure 1.16) as a characteristic
parameter for the ejastic behavior:

s

0
Je
%

(1.50)

Time-dependent Behavior
Viscosity n(t) (Figure 1.17):

ney =28 (1.51)
Yo
Shear compliance J(¢) (Figure 1.18):
ny
T,

0

f’(t)= (1.52)

Maximum time of retardation 7, as a rheological quantity for the transient behavior
(Figure 1.19):

V() = ¥, 11— exp{—t/ T )] (1.53)
g7,
Eé- 7

Figure 1.17  Initial state in a stress test under linear shear flow {21]

/

¢ 4
ng /
e
/ ’z/ 7&
gt —=

Figure 1.18  Initial state in creep under linear shear flow [21]



1.4 Viscoelastic Behavior of Polymers 21

\%

Figure 1.19  Retardation from steady shear flow [21] {shear stress T=0 at time t = 0}

Ig (s 2~
/

§—-—

T et

Figure 1.20 Relaxation after a step shear strain % [21)

[1]
Shear stress relaxation modulus G{(r) (Figure 1.20 and Figure 1.21):

————

}—~

Gy =28 (1.54)
Yo

Dependence of storage modulus G’ and loss modulus G” on frequency (Figure 1.22)
with sinusoidal shear strain ¥

y = ysinot (1.55)
Sinusoidal and out-of-phase shear stress o

7 = 7(G'sinwt + G coswt) (1.56)

where

¥ = amplitude of shear

@ = angular frequency

The storage modulus G'(@) characterizes the elastic behavior, whereas the loss modulus
G” depicts the viscous behavior of the melt subjected to periodic shear deformation.
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lg6{t) —=-
L/

Igi#) —=—
Figure 1.21 Relaxation modulus of linear shear flow [21]

NN

A
ez

Figure 122  Storage modulus G” and loss modulus G” as functions of frequency

A

g —=—

Expressions for conversion:

Determination of zero viscosity and shear compliance from relaxation modulus [21]:

7 = fGuyar (1.57)
= iz [t Gy ar (158)
o o

Determination of zero viscosity and shear compliance from storage and loss moduli
{21]:

G
o = lim = (1.59)
2=Liim& (1.60)
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14.1.2 Nonlinear Viscoelastic Behavior
Steady Shear Flow

The viscosity

%
==
Y

and the shear compliance

L

J. =
€ z

are dependent on the shear rate and shear stress respectively in the nonlinear case. Their
limiting values for small shear rates or shear stresses are 7, and ]E (Figure 1.23). Another
material function in addition to the shear compliance characterizing the elastic behavior
is the primary normal stress coefficient © with N, as the normal stress difference:

o=

%

The limiting value of the normal stress function @(};) (Figure 1.23) at low shear rates is
given by

(1.61)

0, = lim 8(7) (1.62)
Yo —0
In addition, we have:
Jo=— (1.63)
27
% 18]
e
’ Ta -~
% )
2z .
= e .
<+ \ N N\ai)
%} \‘\
~J- 1 u Y \\‘
AY

Figure 1.23  Parameters for steady shear flow [21).| = linear region; Il = nonlinear region
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Characterization of the Transient State
Initial state in a stress test (Figure 1.24)
) < nit) (1.64)

o
n(r) is the asymptote.
Plots of starting state in creep test (Figure 1.25)

72 J¢) (165)
:'(t) is the asymptote.
Relaxation modulus (Figure 1.26):

Git) = h(z) Git) (1.66)

The time-dependent behavior remains unchanged in the nonlinear case. The entire
function plot will be displaced by the factor k(7).

Rl

g7 (£} —

Figure 1.24  Initial state in a stress test of nonltinear shear flow [21]

/ /// )
/

1gf —=

gt} —=

—F

Figure 1.28  Initial state in creep test of nonlinear shear flow [21]
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Gi)

Il
/

/
W/

g&i#) —e=

,
aw

Figure 1.26  Relaxation modulus of nonlinear shear flow [21]

1.4.2 Uniaxial Tension
1.4.2.1 Linear Viscoelastic Behavior
Steady Tensile Extensional Flow

Tensile zero viscosity g, as a material function for the viscous behavior (Figure 1.27 and
Figure 1.28)
%

=20 1.67
Ho Py (1.67)

Steady state tensile compliance D as a material function for elastic behavior:

£,
Do = Gus (1.68)

[+ 0'0

! T TNE
I

g5

> :

} Time # —-

o

g Gy

- i —

= Time # —

Figure 1.27  Time dependence of tensite strain and tensile stress at constant tensile strain rate £,
and subsequent retardation after unloading {tensile stress 7= 0} [21.
&,, = recoverable tensile strain; | = initial state; Il = steady state; lIl = retardation
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-
5
= G
k3
Time #-—=
’ \Ia\fr.s
w ]
g
I - it I} m
1 I
Time § —e=

Figure 1.28 Tensile creep at constant tensile stress 7, and subsequent retardation after unloading
{tensile stress £=0) [211

Transient Behavior
Tensile viscosity ;t(t) (Figure 1.29)

uiey=2% (1.69)
&

) =31146) (1.70)

Hy =37,

The tensile viscosity is three times the shear viscosity.

g
Tensile creep compliance D(t) (Figure 1.30):

Dy = £© (L71)
Ty
D) = % o) (1.72)

Maximum retardation time 7, ,, (Figure 1.31)
£ () = £, [1 — exp(—t/ T )] (1.73)

Relaxation after a step strain of &
[
Tensile relaxation modulus E(t) (Figure 1.32):
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Figure 1.29 Initial state of linear tensile extension [21]

I # —=

v

lgdtH) —=
N\

Figure 1.30 Initial state in tensile creep under linear tensile extension [21]

\
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lplE e —=
/

g

o~

i~

Time § —=-

Figure 1.31  Retardation from steady state tensile extension [21]

“\.“*\

~
AN

gt —=

Figure 1.32  Tensile relaxation modulus of tensile extensional flow [21]

lgE‘(tl-—-
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By =28 (1.74)
£
E¢t) =3 G(t) (1.75)

1.4.2.2 Nonlinear Viscoelastic Behavior

Steady Tensile Extensional Flow

As shown in Figure 1.5, the tensile viscosity g is given by
%%

Tensile compliance D, (Figure 1.33)

£,

_ 1

¥

D

e

195, igJ, —~

3«"%]: : \ \
AN
96, 107, ~

Figure 1.33  Plot of tensile compliance D, and shear compliance J; [21]

Time-dependent Behavior
Plots of starting state in tension test (Figure 1.34) [21]

26 2 u(e) (1.76)

H(t) is an asymptote.
Start-up curves in a tensile creep test {Figure 1.35} [21]

D(t) < D(¢) (1.77)

4]
D({t) is an asymptote.
Tensile relaxation modultus (Figure 1.36)

Er,2,) < gleo) - E() (1.78)
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Figure 1.34 Initial state of nonlinear tensile extension [21]
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s

g2t} ——

Figure 1.35 Initial state of tensile creep under nonlinear extension [21]

1 “\\.‘Qs ”
ht

Eq

s

Figure 1.36 Tensile relaxation modulus as a function of time under nonlinear tensile extension [21]

1.4.3 Maxwell Model

The viscoelastic properties of a polymer can be used to calculate the deformation of a
bubble in a film blowing process. In this case, the Maxwell model, which defines a

viscoelastic body as an ideal spring and a dashpot in series (Figure 1.37) can be applied
[18,19].
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Figure 1.37  Maxwell fluid [1]

The total rate of deformation 7 is the sum of the elastic component #; and viscous
component iy

7= Fu + i (1.79)
leading to [22]

MF=tt+7 (1.80)
where

f, = gN_H (1.81)
The time ¢, is called the relaxation time, as the stress 7 relaxes with the time.

1}y = Newtonian viscosity
Gy = elastic shear modulus of the spring {Hooke element)

At a given rate of deformation, the viscosity 77 5 reaches the Newtonian value asympto-
tically (Figure 1.38). After the release of strain the stress delays according to

T =7, exp(—t/t,) {1.82)

It can be seen from Equation 1,82 that the relaxation time is the period, in which the
stress decreases to 1/e (37%) of its original value [23]. It also follows from Equation 1.82
that the modulus of relaxation G for y= ¥, is given by

G =G, exp(~t/t,) (1.83)

T

Mg —o—
,

1
t
I
)
/
I h
” 1
I
1
|

BL

}

Figure 1.38  Tensile viscosity of Maxwell fiuid {1}
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The expression for the elongation of the bubble in a film blowing process can now be
given as
1o dv

il (1.84)

+t —=
¢ S or dx

where

O = tensile stress,

. = relaxation time,

H = tensile viscosity of the melt,

v = vertical velocity component of the bubble,
x = axial coordinate.

~

As the elongation of the bubble occurs biaxially, the deformation in the circumferential
direction has to be calculated on similar lines. Assuming that the tensile viscosities and
the relaxation times in both directions are equal, the influence of viscoelasticity on the
bubble form can be predicted [18]. WaGNER [18] estimates the relaxation time in the
order of 5 to 11 s, depending on the operating conditions.

1.4.4 Practical Formulas for Die Swell and Extensional Flow

Elastic effects are responsible for die swell, which occurs when the melt exits through a
die as shown in Figure 1.39 [20].

The following equation for the die swell is given by CogswerL [24]:

B z_i_,,k (H_L]z 1 (1.85)

where

By = dieswell 4 (Figure 1.39) in a capillary with a length-to-diameter ratio
d greater than 16 and
Y% = recoverable shear strain

In Figure 1.40, the recoverable shear strain is presented as a function of die swell according

to Equation 1.85 [24]. s

&

Figure 1.39  Die swell in extrusion



32 1 Formulas of Rheology {Refs. on p. 33]
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Die swell 8 —=
Figure 1.40  Dependence of recoverable shear on die swell [24]
Extensional Flow

The following relationships for extensional flow of melt after CocsweiL [24] are important
in practice:

Elongational stress o:

3
O = E("R +1}- p, (1.86)
Tensile viscosity p:
2 2
#3Q["—] 187
2 A
Reversible extension &
& =B} (1.88)
Rupture stress oy
3
On =§(_’mR +1} p. (1.89)
where
np  =reciprocal of power law index n in Equation 1.28
p. = entrance pressure loss according to Equation 1.20

%>, = apparent shear rate and apparent shear viscosity, respectively
By, = die swell in melt flow through an orifice with zero length
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2 Thermodynamic Properties of Polymers

In addition to the rheological data, thermodynamic properties of polymers are necessary
for designing machines and dies. It is often the case that the local values of these data are
required as functions of temperature and pressure. Besides physical relationships, this
chapter presents regression equations developed from experimental data for calculating
thermodynamic properties, as these polynomials are often used in the practice, for
example, in data acquisition for data banks [1].

2.1 Specific Volume

According to the Spencer Gilmore equation, which is similar to the van der Waal equation
for real gases, the relationship between pressure p, specific volure v, and temperature T
of a polymer can be written as

. +» RT
(v-b)(p+p) W (2.1)
In this equation, b is the specific individual volume of the macromolecule, p’ is the
cohesion pressure, W is the molecular weight of the monomer, and R is the universal gas
constant [2].

Example

Following values are given for a LDPE:
W =28.1 g/Mol;
b =0.875cm’/g;
p =3240 atm

Calculate the specific volume vat T=190°C and p = 1 bar.

Solution

Using Equation 2.1 and the conversion factors to obtain the volume v in cm?/g, we obtain

v = 10 8.314 x (273 + 190)
28.1x3240.99x1.013

The density p is the reciprocal value of the specific volume so that

+0.875 = 1.292 cm’/g

p=_ (2.2)

v
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Figure 2.1 Specific volume as a function of temperature and pressure for LOPE [1]

The functional relationship between specific volume v, pressure p, and temperature T
can also be expressed by a polynomiat of the form {1, 3]
v=A0), + AQl), - p+ AQ), -T+A(3),-T-p (2.3)

if experimental data are available (Figure 2.1). The empirical coefficients A(0), ... A(3),
can be determined by means of the computer program given in [1].

2.2 Specific Heat

The specific heat ¢, is defined as
oh
=|— 2.4
-~ aT)P 29)
where

h = Enthalpy
T = Temperature

The specific heat ¢, defines the amount of heat that is supplied to a system in a reversible
process at a constant pressure to increase the temperature of the substance by dT.

The specific heat at constant volume c, is given by

(&
¢, = (BT ]v (2.5)
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where

© = internal energy
T = temperature

In the case of ¢, the supply of heat to the system occurs at constant volume.
¢, and ¢, are related to each other through the Spencer-Gilmore equation (Equation 2.1):
R
£, = CP - W (2'6)
The numerical values of ¢, and ¢, differ by roughly 10%, so that for approximate calcu-
lations ¢, can be considered equal to ¢, [2].

Plots of ¢, as a function of temperature are shown in Figure 2.2 for amorphous,
semicrystalline, and crystalline polymers.

As shown in Figure 2.3, the measured values can be expressed by a polynomial of the
type
co(T) = A®)c, + Al) ¢, - T + AQ)c, - T? (2.7)

a} b) c)

-

i h

Figure 2,2 Specific heat as a function of temperature for amorphous (a), semi crystalfine (b),
and crystalline polymers (c) [4]
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£ =7
A 23
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Figure 2.3  Comparison between measured values of ¢, [6] and polynomial for LDPE (1]
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The use of thermal properties ¢, and p in design problems is illustrated in the examples
given in Chapter 6.

The expansion coefficient ¢, at constant pressure is given by (4]

1{ ov
a, =—| — 2.8
oy [aT Jp (2.8)
The linear expansion coefficient oy, is approximately
1
&y = g o, (2.9)

The isothermal compression coefficient ¥ is defined as [4]

1{ ov
=== 2.10
Yk . ( Py JT (2.10)
a, and % are related to each other by the expression [4)
cp=cv+L'a2" (2.11)
Yk
2.3 Enthalpy
Equation 2.4 leads to
dh=c,-dT (2.12)

As shown in Figure 2.4, the measured data on h = h(T) [6] for a polymer melt can be
expressed by the polynomial

R(T) = A(O), + AQD), - T + AQQ), - T? {2.13)

The specific enthalpy defined as the total energy supplied to the polymer divided by
throughput of the polymer is a useful parameter for designing extrusion and injection
molding equipment such as screws. It provides the theoretical amount of energy required
to bring the solid polymer to the process temperature.

Values of this parameter for different polymers are given in Figure 2.5 [4].

If, for example, the throughput of an extruder is 100 kg/h of polyamide (PA) and the
processing temperature is 260 °C, the theoretical power requirement would be 20 kW.
This can be assumed to be a safe design value for the motor horse power, although
theoretically it includes the power supply to the polymer by the heater bands of the
extruder as well.
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24  Thermal Conductivity

The thermal conductivity A is defined as

R AL N (2.14)
t-A-(T,-T,)
where
Q = heat flow through the surface of area A in a period of time ¢

(T,-T,) = temperature difference over the length !

Analogous to the specific heat ¢, and enthalpy 4, the thermal conductivity A can be
expressed as [1]

AMT) = A0); + AQ), - T+ AQR), - T? (2.15)

as shown in Figure 2.6.

=3
(¥

e measured
polynari?-‘“""--..__

2=

Thermat conduclivity A —e=
=

£
L=

30 100 150 o 200

Temperaiure [ —=-
Figure 2.6 Comparison between measured values of A [6] and polynomial for PP [1]

The thermal conductivity increases only slightly with pressure. A pressure increase from
1 bar to 250 bar leads to an increase of only less than 5% of its value at 1 bar.

Within a particular resin category such as LDPE, HDPE, the thermal properties are largely
independent of the molecular structure. Exhaustive measured data of the quantities ¢,
h, and A and pressure-volume-temperature diagrams of polymers are given in the VDM A-
Handbook [5].

Approximate values of thermal properties useful for plastics engineers are summarized
in Table 2.1 [4].
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Table 2.1  Approximate Values for the Thermal Properties of Selected Polymers [4]

Polymer Thermal Specific heat Density Glass transition Melting

conductivity temperature point range

A ¢ P T, T,

Wim-K ki/kg K glam® °C °C

PS 0,12 1.20 1.06 101 -

PVC 0.21 1.10 1.40 80 -

PMMA 0.20 1.45 1.18 105 -

SAN 0.12 1.40 1.08 115 -

ABS 0.25 1.40 1.02 115 -

PC 0.19 1.40 1.20 150 -
LDPE 0.24 2.30 0.92 —120/-50 ca. 110
LLDPE 0,24 230 092 —120/-90 ca. 125
HDPE 0.25 225 0.95 —-120/-90 ca. 130
PP 0.15 2.10 0.91 -10 160~-170
PA-6 0.25 2.15 1.13 50 215-225
PA-6.6 0.24 2.15 1.14 55 250260
PET 0,29 1.55 1.35 70 250260
PBT 0.21 1.25 1.35 45 ca. 220
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3 Formulas of Heat Transfer

Heat transfer and flow processes occur in the majority of polymer processing machinery
and often: determine the production rate. Designing and optimizing machine elements
and processes therefore require knowledge of the fundamentals of these phenomena.
The flow behavior of polymer melts has been dealt with in Chapter 2. In the present
chapter, the principles of heat transfer relevant to polymer processing are treated and
explained with examples.

3.1 Steady State Conduction

Fourier’s law for one-dimensional conduction is given by

O=-4 A% (3.1)

where

Q = heat flow thermal conductivity

A = area perpendicular to the direction of heat flow
T = temperature

x = distance (Figure 3.1)

Figure 3.1 Plane wall [1]

3.1.1 Plane Wall
Temperature profile (Figure 3.1) [1]:

T = e~ Fon)

c x4+ Twl (3.2)
Heat flow:

Q=2 A-(Ty - Ti) (33)
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Analogous to Ohm’s law in electric circuit theory, Equation 3.3 can be written as [2]

. AT
=— 3.4
Q 2 (3.4)
in which
d
R=— 3.5
1A (3.5)
where

AT = temperature difference
¢ = wall thickness

R = thermal resistance
Example

The temperatures of a plastic sheet {36 mm thick} with a thermal conductivity A =
0.335 W/(mK)are T, =100°C and T;,, = 40 °C according to Figure 3.1. Calculate the
heat flow per unit area of the sheet.

Solution
Substituting the given values in Equation 3.3 we obtain

Q__03% | (100 — 40) = 670 W/m*

A (30/1000)

3.1.2 Cylinder
Temperature distribution (Figure 3.2) {1}:

Ty, — T
T(r) = Ty, + —2 0. 1n(-’l] (3.6)

Figure 3.2 Cylindrical wall [1]
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Heat flow:
y=2.4 @, -1, 3.7)
Q_E' m‘(Wl-Wz) (°
with the log mean surface area A, of the cylinder
A, = Ay ; A (3.8)
ln(—z]
Al
where §=r,-r,.
3.1.3  Hollow Sphere
Temperature distribution [1]:
Tw — T ) Tty — T B
T(r)zl( w, ~ Tw, )1 2_( wih = Ty, 1) (3.9)
r nmh h=n
with the boundary conditions
Tr=n)= TW1 and T(r=n)= T‘,‘,,2
Heat flow:
y=2 Ty — T; 3.10
Q=5 An (T~ Ti) (3.10)
The geometrical mean area A, of the sphere is
A, =+ A 4, (3.11)
The wall thickness dis
S=1n-1 (3.12)
3.1.4 Sphere
Heat flow from a sphere in an infinite medium (r, — =) {1]
Q=471 1 ATy - T.) (3.13)

where T, = temperature at a very large distance.

Figure 3.3 shows the temperature profiles of the one-dimensional bodies treated above

[1].
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Figure 3.3

3.1.5

f

One-dimensional heat transfer [1} a: sphere, b: cylinder, c: plate

Heat Conduction in Composite Walls

Following the electrical analogy, heat conduction through a multiple layer wall can be
treated as a current flowing through resistances connected in series. With this concept
we obtain for the heat flow through the composite wall as shown in Figure 3.4

o = T, =Q/‘151A
]

. 0

TWI_TW3 :(;)Azi1
2

. 0

ij_-TW4 :Qﬂa;
3

(3.14)

(3.15)

(3.16)

Adding Equation 3.14 to Equation 3.16 and setting A, = A, = A, = A gives

TW_’W

Figure 3.4

1

a4 L 6 &
*‘Q[AIA+)QA+}L_,3AJ_AT

Heat transfer through a compaosite wall [1]

(3.17)
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Thus
Q= ar (3.18)
i.,6 5
AA LA LA
Inserting the conduction resistances
ol
R = 3.19
=T A (3.19)
d.
R, = 3.20
"4, .
8,
Ry = —2 3.21
S A Ay 2
into Equation 3.18 we get
Q= ___AT _AT (3.22)
(R +R,+R;) R
Example 1

A two-layer wall consists of the following insulating materials (see Figure 3.4);

d,=140mm, 1, =0.033 W/(mK)

The temperatures are Ty, =30°C, T, =2°C. Calculate the heat loss per unit area of
the wall.

Solution

AT =Ty Ty, =30-2=28°C

Area A=1m’
RO 1671000 .y
T A4A 00481
R, = 8, _140/1000 _ . vny
A A 00331
Q= AT 28 6.13W

(R, +R;) (424+033)

The following example [2] illustrates the calculation of heat transfer through a tube as
shown in Figure 3.5.
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Figure 3.5 Heat flow in a multilayered cylinder

Example 2
A tube with an outside diameter of 60 mm is insulated with the following materials:

d,=50mm, I, =0.055W/(mK)
d,=40mm, I, =005 W/ (mK)

The temperatures are Ty =150°C and Ty, =30 °C. Calculate the heat loss per unit
length of the tube.
Solution

Resistance R,:

R B _ 0.05
"TAA 0055-27F-L

average radius 7 :

%= 80-30) _830) = 50.97 mm
)
30
R = 2.839 K/W
L
Resistance R,:
8 0.04

R, = L =

T LA, 005-27%-L
average radius % :
_ (120 - 80)

[120] = 98.64 mm
In| —
80

h
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43 & X

Figure 3.6 Heat transfer in composite walls in paraliel 11}

Heat loss per unit length of the tube according to Equation 3.22:

Q__150-30
L 2.839+1.29]1
In the case of multiple-layer walls, in which the heat flow is divided into parallel flows as

shown in Figure 3.6, the total heat flow is the sum of the individual heat flows. Therefore
we have

=29.1 W/m

Q=Y (3.23)
i=1

i T, —T,

Q=>4 A;)% (3.24)
=]

3.1.6 Overall Heat Transfer through Composite Walls
If heat exchange takes place between a fluid and a wall as shown in Figure 3.7, in addition
to the conduction resistance we also have convection resistance, which can be written as
_ 1
oA

where a7 = heat transfer coefficient in the boundary layer near the walls adjacent to the
fluids.

R, (3.25)

R

ay [+ £
Fiuid 1 _ Fluid 2
A'l - X N Y _Az

-

Figure 3.7 Conduction and convection through a composite wall [1]
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The combination of convection and conduction in stationary walls is called overall heat
transfer and can be expressed as

Q=kA -T)=— (T, -T) (3.26)
Ry,

where k is denoted as the overall heat transfer-coefficient with the corresponding overall
resistance Ry,

1
Ry =—
YA

Analogous to conduction for the composite wall in Figure 3.7, the overall resistance Ry,
can be given by

(3.27)

Ry=R, +X R +R, (3.28)
i=l
or
1 z
3.29
kA o) Al ; 0’2 Az G2)

A simplified form of Equation 3.29 is

k= ! (3.30)

1 = 3. 1
-+ _*+_
2 E& )

Calculation of the convection heat transfer coefficient is shown in the Section 3.5.

3.2 Transient State Conduction

The differential equation for the transient one-dimensional conduction after Fourier is
given by

2
%—T= ag% (3.31)

where

T = temperature
t =time
x = distance
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The thermal diffusivity « in this equation is defined as
A

-~ (3.32)
pe

il =

where

A = thermal conductivity
¢, = specific heat at constant pressure

p =density

The numerical solution of Equation 3.31 is given in Section 4.3.4. For commonly
occurring geometrical shapes, analytical expressions for transient conduction are given
in the following sections.

3.2.1 Temperature Distribution in One-Dimensional Solids
The expression for the heating or cooling of an infinite plate [2] follows from Equation
3.31 (Figure 3.8):
Tw-Tp _ 8 [e"" oy Lyomp 1 osan J (3.33)
Tw-T, o 25
The Fourier number F, is defined by

F = ’ {3.34)
where

T,,= constant surface temperature of the plate
T, = imitial temperature

T, = average temperature of the plate at time £
t, = heating or cooling time s

X = half thickness of the plate [ X = = ]

a, =(m/2) 2

a = thermal diffusivity, Equation 3.32

7

7z
A
7
7

Figure 3.8 Non-steady-state conduction in an infinite plate
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The equation for an infinite cylinder with the radius r_ is given by {2]
Ty~
Tw - T,

and for a sphere with the radius r,

=0.692¢778B 10,1317 100534745 4, (3.35)

Tw =T _ 6608295 1015265 1006766 %5% 4 (3.36)
Ty - T,
where
.t
Ry =2k (3.37)
rl‘ﬂ

In the range of F;, > 1, only the first term of these equations is significant. Therefore, for
the heating or cooling time we obtain {2]

Plate:
2
t = 1-(% 1{[%) [MH (3.38)
Cylinder
LT e
fo=—m_1n| 0602 = Ts (3.39)
578a | Ty - T
Sphere
. v
f= - _in| o608 ke (3.40)
9.870 w -Tb ]

The solutions of Equatlon 3.32 to Equation 3.37 are presented in a semi-logarithmic
plot in Figure 3.9, in which the temperature ratio 8— = (Tyy - T,) /(T - T,)is shown
as a function of the Fourier number F,.

Not considering small Fourier numbers, these plots are straight lines approximated by
Equation 3.38 to Equation 3.40.

If the time # is based on the centre line temperature T, instead of the average temperature
T,,» we get [3]

2 ——
o= in| 4wk (3.41)

Analogous to Figure 3.9, the ratioc @7 with the centre line temperature T;, at time #, is
plotted in Figure 3.10 over the Fourie? number for bodies of different geometry [4].
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Figure 3,9 Average temperature of an infinite slab (c), a long cylinder (k) and a sphere {a) during
non-steady heating of cooling [2]
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The foregoing equations apply to cases, in which the thermal resistance between the
body and the surroundings is negligibly small { o, — <}, for example, in injection molding
between the part and the coolant. This means that the Biot number should be very large,
Bi — co. The Biot number for a plate is

a, - X

Bipu, = = (3.42)

where

¢, = heat transfer coefficient of the fluid
A = thermal conductivity of the plastic

As the heat transfer coefficient has a finite value in practice, the temperature ratio 91‘.,
based on the centre line temperature, is given in Figure 3.11 as a function of the Fourier
number with the reciprocal of the Biot number as parameter [5].

Example 1 [6]

Cooling of a part in an injection meold for the following conditions:

Resin: LDPE
Thickness of the plate: s =127 mm

Temperature of the melt: T, =243.3°C
Mold temperature: Tw =21.1°C
Demolding temperature: T, =76.7 °C
Thermal diffusivity: a =129107 cm?s

The cooling time #,_ is to be calculated.

Solution

The temperature ratio Oy, :

0, = —Tw _ 767-211 =025
b Tw -T, 2433-21.1
Fourier number F, from Figure 3.10 at ©5 = 0.25
Fo = 0.65

cooling time §:

X =

—
Nl!"-'
]

2 = 6.35mm
2
a-t,  129-107° -4

= = =0.65
x? 0.635°
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Example 2 {6]

Calculate the cooling time #, in Example 1 if the mold is cooled by a coolant with a heat
transfer coefficient of ¢, = 2839 W/(m” - K).

Solution

e, = 2839 W/(m* - K)
Aplastic =0.242 W}"(m . K)

The resulting Biot number is

pi=f X _28963 - 0 L_oooie
A 0.242-1000 Bi
As can be extrapolated from Figure 3.11, the Fourier number does not differ much from
the one in the pre\rlous example for © = 0.25 and 1/Bi =0.01342. The resistance due
to convection is therefore negligible and the cooling time remains almost the same.
However, the convection resistance has to be taken into account in the case of a film with
a thickness of 127 g that is cooling in quiescent air, as shown in the following calculation:

The heat transfer coefficient for this case is approximately
a, = 5678 W/(m? - K)
The Biot number Bi

TS5 oo Log
10° - 0.242 Bi
F, from Figure 3.11
'Fl] =95

The cooling time

X* F, 655 .10°-95

5 =297s
a 10° - 1.29

tk=

Example 3 [7]
Cooling of an extruded wire

A polyacetal wire of diameter 3.2 mm is extruded at 190 °C into a water bath at 20 °C,
Calculate the length of the water bath to cool the wire from 190 °C to a centie line
temperature of 140 °C. The following conditions are given:

a, = 1700 Wr‘(mz-K}
i = 107 m%fs
zphsm 0.23 W/(m - K)

hanl-off rate of the wire Vi; = 0.5 m/s
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Solution
a,- X

The Biot number Bi =

where R = radius of the wire

L S T i = 0.0846

1000 - 0.23 ' Bi

The temperature ratio O

8, - T, - Ty _140-20 _ 0.706
® Tw-T, 190-20
The Fourier number F, for 8y, = 0.706 and —L = 0.0846 from Figure 3.11 is approxi-
mately Bi

Fy=0.16
The cooling time §, follows from

a- tk = 10_? tk tk

=0.16
R (6 10‘3‘)2 2.56-10

=41s
The length of the water bath is

Vg £ =05-41=205m

3.2.2 Thermal Contact Temperature

If two semi infinite bodies of different initial temperatures @, and ©,, are brought
into contact as indicated in Figure 3.12, the resulting contact temperature Oy is given

by (3
RULrENN
O = — KErn) (3.43)
(J/?.pc)z

(\I Apeh

where

A = thermal conductivity
fol = density

¢ = gpecific heat

J A pc = coefficient of heat penetration
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Figure 3.12 Temperature distribution in semi infinite solids in contact [3]

Equation 3.43 also applies for the case of contact of short duration between thick bodies.
It follows from this equation that the contact temperature depends on the ratio of the
coefficients of heat penetration and lies nearer to the initial temperature of body that has
a higher coefficient of penetration. The ratio of the temperature differences (0, — )
and (B — ©,,) are inversely proportional to the coefficient of penetration:

81"!.1 _QK — (\J A’pc)z (3.44)
eK—'e_AZ (Jj.pc)]_

Example

According to Equation 3.43 (8] the contact temperature @, of the wall of an Injection
mold at the time of injection is

by ©,  +h, 0,

&) = 3.45
W max bw + bp ( )
where
b =.JApc
©®,, . = temperature before injection

2] Mmm = melt temperature
Indices w and p refer to mold and polymer, respectively.

As shown in Table 3.1 [8], the coefficients of heat penetration of metals are much higher
than those of polymer melts. Hence the contact temperature lies in the vicinity of the
mold wall temperature before injection.

The values given in the Table 3.1 refer to the following units of the properties:

thermal conductivity : W/{(m - K)

density o kg/m’
specific heat & kJ/(kg - K)
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Table 3.1 Coefficients of Heat Penetration of Mold Material and Resin [8)

Material Coefficient of heat penetration b
wsms R m—a R K_I

Beryllium copper {BeCu 25} 17.2-10

Unalloyed steel (C45W3) 13.8-10°

Chromium steel (X40Crl3) 1.7 10’

Polyethytene (HDPE) 0.9 10’

Polystyrene (PS) 0.57 - 10°

The approximate values for steel are

A =50 W/(m-K)
p = 7850 kg/m’
¢ =0485kJ/(kg-K)

The coefficient of heat penetration b

b=JA-p-c=450.7.85-10" - 0.485-10° =13.8-10° Ws** - m™> . K’

33 Heat Conduction with Dissipation

The power created by the tangential forces in the control volume of the fluid flow is
denoted as dissipation [9]. In shear flow the rate of energy dissipation per unit volume is
equal to the product of shear force and shear rate [10]. The power due to dissipation [11]
therefore is:

Ei=1-% (3.46)
From the power law we get
L
Ey = [iJ g (3.47)
d K 4
For a Newtonian fluid with # = 1 we obtain

The applicable differential equation for a melt flow between two plates, where the upper
plate is moving with a velocity U, (Figure 2.4) and the lower plate is stationary [11] is

2 2
Ag-]:m(a“*] =0 (3.49)
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For drag flow the velocity gradient is given by

o _Uy (3.50)
dy H
Equation 3.49 can now be written as
2 z
,13_12"+q[£] -0 (3.51)
oy H

If the temperature of the upper plate is T, and that of lower plate is Ty, the temperature
profile of the melt is obtained by integrating Equation 3.51. The resulting expression is

nu, -y ¥y y
T="x Y X e LT+ T 3.52
ZAH( H] H(‘ o)+ T (3-52)

As shown in Section 4.2.3, this equation can be used to calculate the temperature of the
melt film in an extruder.

3.4 Dimensionless Groups

Dimenstonless groups can be used to describe complicated processes that are influenced
by a large number of variables with the advantage that the entire process can be analyzed
on a sound basis by means of a few dimensionless parameters. Their use in correlating
experimental data and in scaling-up of equipment is well known.

Table 3.2 shows some of the dimensionless groups often used in plastics engineering.

Table3.2 Dimensionless Groups

Symbol Name Definition
Bi Biot number ali A

Br Brinkman number nw [ {AAT)
Deb Deborah number t. /e,

F, Fourier number atlf

Gr Grashof number gB-ATE IV
Gz Graetz number Pia:s)

Le Lewis number al é

Na Nahme number Bwnii
Nu Nusselt number alt A

Pe Peclet number wlia

Pr Prandtl number via

Re Reynolds number pwlin

Sh Sherwood number Biré

Sc Schmidt number vié

sk Stokes number P-lfin-w)
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Nomenclature:

a: thermal diffusivity {(m?fs)
g acceleration due to gravity (m/s?)
I characteristic length {m)

p: ressure (N/m?)
£ time (s)
Indices

D,P: memory and process of polymer respectively

AT: Temperature difference (K)

Velocity of flow (m/s)

Outside heat transfer coefficient [W/(m? - K}]

Coefficient of volumetric expansion ( K

Temperature coefficient in the power law of viscosity (K™')
Mass transfer coefficient {m/s)

Diffusion coefficient {m*/s)

Viscosity (Ns/m?)

Thermal conductivity (Index i refers to the inside valune) [W/(m K)]
Kinematic viscosity (m?/s)

Residence time (s)

Density (kg/m>)

ROI P OPPRR E

3.4.1  Physical Meaning of Dimensionless Groups

Biot number: Ratio of thermal resistances in series: (I/ A;)) / (1/ o)
Application: heating or cooling of solids by heat transfer through conduction and
convection

Brinkmann number: ratio of heat dissipated (1 wz) to heat conducted (A AT)
Application: polymer melt flow

Fourier number: ratio of a characteristic body dimension to an approximate temperature
wave penetration depth for a given time [16}
Application: unsteady state heat conduction

Deborah number: ratio of the period of memory of the polymer to the duration of
processing [13]. At Deb > 1 the process is determined by the elasticity of the material,
whereas at Deb < 1 the viscous behavior of the pelymer influences the process remarkably.
Grashof number: ratio of the buoyant force g B+ AT P to frictional force (v)
Application: heat transfer by free convection

Graetz number: ratio of the time to reach thermal equilibrium perpendicular to the
flow direction (F/a) to the residence time (t,)
Application: heat transfer to fluids in motion
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Lewis number: ratio of thermal diffusivity {(a) to the diffusion coefficient (&)
Application: phenomena with simultaneous heat and mass transfer.

Nusselt number: ratio of the total heat transferred (¢t - 1) to the heat by conduction ()
Application: convective heat transfer.

Peclet number: ratio of heat transfer by convection (p ¢, - w - 1) to the heat by conduction
W)

Application: heat transfer by forced convection.

Nahme or Griffith number: ratio of viscous dissipation (B wz‘q) to the heat by
conduction (A) perpendicular to the direction of flow
Application: heat transfer in melt flow

Prandtl number: ratio of the kinematic viscosity (v) to thermal diffusivity (a)
Application: convective heat transfer

Reynolds number: ratio of the inertial force {p w I} to viscous force (1)

Application: The Reynolds number serves as a criterium to judge the type of flow. In
pipe flow, when Re is less than 2300 the flow is laminar. The flow is turbulent at Re
greater than about 4000. Between 2100 and 4000 the flow may be laminar or turbulent
depending on conditions at the entrance of the tube and on the distance from the entrance
[2]

Application: fluid flow and heat transfer.

Sherwood number: ratio of the resistance to diffusion (I/ &) to the resistance to mass
transfer (1/ f3,)
Application; mass transfer problems

Schmidt number: ratio of kinematic viscosity (v) to the diffusion coefficient ()
Application: heat and mass transfer problems

Stokes number: ratio of pressure forces (p - I) to viscous forces (17 - w)

Application: pressure flow of viscous media like polymer melts.

The use of dimensionless numbers in calculating non Newtonian flow problems is
illustrated in Section 4.3.3 with an example,

3.5 Heat Transfer by Convection

Heat transfer by convection, particularly by forced convection, plays an important role
in many polymer processing operations such as in cooling a blown film or a part in an
injection mold, to mention only two examples. A number of expressions can be found in
the literature on heat transfer [3] for calculating the heat transfer coefficient & (see
Section 3.1.6). The general relationship for forced convection has the form

Nu = f(Re,Pr) (3.53)
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The equation for the turbulent flow in a tube is given by [16]

Nu = 0.027 Re*®. Pr (3.54)
where

1t = 0.4 for heating
n = 0.3 for cooling

The following equation applies to laminar flow in a tube with a constant wall temperature

(3]

Nuyn = 4 3.66° + 1.61° Re - Pr -, /1 (3.55)
where
d, =inside tube diameter
I =tube length

The expression for the laminar flow heat transfer to flat plate is [3]

Nu,,, =0.664 \ Re 3 Pr (3.56)
Equation 3.56 is valid for Pr = 0.6 to 2000 and Re < 10°.

The equation for turbulent flow heat transfer to flat plate is given as {3]

0.037 Re®®. Pr
1+ 2.448 Re™*Y(Pr?*- 1)

Nutm.b = (3.57)

Equation 3.57 applies for the conditions:
Pr=0.6102000and 5 - 10° <Re < 10.
The properties of the fluids in the equations above are to be found at a mean fluid
temperature,
Example

A flat film is moving in a coating equipment at a velocity of 130 m/min on rolis that are
200 mm apart. Calculate the heat transfer coefficient c¢if the surrounding medium is air
at a temperature of 50 °C.

Solution

The properties of air at 50 °C are:

Kinematic viscosity v =17.86-10° m%/s
Thermal conductivity A =28.22- 107 W/(m - K)
Prandt] number Pr =0.69
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The Reynelds number Re;, based on the length L = 200 mm is
1 -
Re; =130- e 0.2/17.86 - 107" = 24262

Substituting Re; = 24262 and Pr = 0.69 into Equation 3.56 gives
Nuy,, = 0.664 - 24262° - 0.69'° = 91,53

As the fluid is in motion on both sides of the film, the Nusselt number is calculated
according to [3]

Nu = f Nuj,,, + NuZ,,

For turbulent flow Nu,,,, follows from Equation 3.57:

0.037 - 24262%% . 0.69
Nu, , = =102
b T 4 2.448 - 24262701 (0.697 — 1)

The resulting Nusselt number Nu is

Nu = J 91.53% + 1022 =137

Heat transfer coefficient o results from

g Nu-d_ 137 - 2822 -10°°

=1933 W/{m® -K
i 0.2 Am®-K)

3.6  Heat Transfer by Radiation

Heating by radiation is used in thermoforming processes to heat sheets or films, so that
the shaping process can take place. Because at temperatures above 300 °C a substantial
part of the thermal radiation consists of wavelengths in the infrared range, heat transfer
by radiation is also termed as infrared radiation [14]. According to the Stefan-Boltzmann
law the rate of energy radiated by a black body per unit area ¢, is proportional to the
absolute temperature T to the fourth power (Figure 3.13) [1]:

¢, =0T (3.58)
The Stefan-Boltzmann constant has the value
o= 5771072 W/(cm? - K*)
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Figure 3,13 Figure 3,14 Figure 3.15
Black body radiation [1) Lambert’s law [1] Properties of radiation

Equation 3.58 can also be written as

: TY
& =, [ﬁ) (3.59)

where ¢, = 5.77 W/ (m?-KY

The dependence of the black body radiation on the direction (Figure 3.14) [1] is given
by the cosine law of Lambert

é, = ¢, cosg (3.60)
The radiation properties of technical surfaces are defined as (Figure 3.15) {1]:

Reflectivity p= % (3.61)
Absorptivity a= % (3.62)
Transmissivity d= % (3.63)

The sum of these fractions must be unity, or
pra+d=1
The transmissivity & of opaque solids is zero so that
proa=1
The reflectivity of gases p is zero and for those gases which emit and absorb radiation
a+d=1
Real bodies emit only a fraction of the radiant energy that is emitted by a black body at
the same temperatare. This ratio is defined as the emissivity £ of the body,

= (3.64)
eS



66 3 Formulas of Heat Transfer [Refs. on p. 71]

At thermal equilibrium according to Kirchhoff s identity
e=o (3.65)
Radiation heat transfer between nonblack surfaces,

The net rate of radiant heat exchange between two infinite parallel plates is given by [15]
: Y (nY

=Ag =L - —2) 3.66

Q; 12 6 [(100) [100 j| (3.66)

A =area
£,, = emissivity factor and is defined by

where

£y = (3.67)
1 1
—4—=1
& &
Indices 1 and 2 refer to the two plates,
When T, is much smaller than T, the heat flow is approximately
T 4
=8¢ Al =L 3.68
Q= a6 4( ) 669

When the heat transfer takes place by radiation and convection, the total heat transfer
coefficient can be written as [15]

Qiotal = Xeonvection T Fradiation
where

Qs
& - = —_——
radiation A (T‘.i _ Tz)

Example

A plastic sheet moving at a speed of 6 m/min is heated by two high-temperature heating
elements. Calculate the power required for heating the sheet from 20 °C to 140 °C:

net enthalpy of the plastic for a temperature difference of 120 °C: Ak = 70 k)/kg

Width of the sheet w =600 mm
Thickness s =250
Density of the resin p =900 kg/m’

Area of the heating element A = 0.0093 m’
Emissivity of the heater =09
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Solution

Heating power Ny

Mass flow rate of the plastic #t:

6- 0.6 250 900
10°

NH=mM=O.81-76f;-1000=945W

As the area of the heating element is small compared to that of the sheet the equation
applies [14]

4
é=E€c, [lJ W/m’

m=U.-w-s5-p= = 0.81 kg/min

100

total area of the heating element A, = 2 - A so that we have

¢ Ay =Ny
T 4

0.95.77 2 0.0093 - [—) = 945
100

L 995
100

T=995K

3.7 Dielectric Heating

The dielectric heat loss that occurs in materials of low electrical conductivity when placed
inan electric field alternating at high frequency is used in bonding operations, for example,
to heat-seal plastic sheets or films.

The power dissipated in the polymer is given by [14]

Ny=2xf-C-E%-cotg (3.69)
where

Ny;= power (W)

f = frequency of the alternating field (s™)
C = capacitance of the polymer (farads)

E = applied voltage {Volt)

¢ = phase angle
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The rate of heat generation in a plastic film can be obtained from Equation 3.69 and
given as [15]

Et-f-&
Ny =557 ————=+ 3.70
" { 45 ] (3.70)

where

Ny = rate of heat generation (Wim?)
g = dielectric loss factor
b = half thickness of the film (n)

Example
Given:
E =300V
f =10MHz
£ =024
b =50

Calculate the rate of heat generation and the time required to heat the polymer from
20°C to 150 °C.

Substituting the given values in Equation 3.70 gives

_55.7-500” -10-10° - 0.24

> = 3.34-10° W/m®
4.50

Ny

The maximum heating rate AT per second is calculated from
Ny
P

AT = (3.71)

For
Ny = 3.39- 109 W/m?
¢ =22 kjf (kg - K)
p = 800 kg/m’

_3.39-10°

AT =
800-2.2

= 1926 K/s

Finally the heating time is

150 - 20
1926

= 0.067 s




3.8 Fick’s Law of Diffusion 69

3.8 Fick’s Law of Diffusion

Analogous to Fourier’s law of heat conduction (Equation 3.1) and the equation for shear
stress in shear flow, the diffusion rate in mass transfer is given by Fick’s law. This can be
written as [16]

where

m, = mass flux per unit time

A  =Area

D,y = diffusion coefficient of the constituent A in constituent B
¢, = Inass concentration of component A per unit volume

x = distance.

The governing expression for the transient rate of diffusion is [2)

Jc %
Tere o7
x
where
t = time
x = distance

The desorption of volatile or gaseous components from a molten polymer in an extruder
can be calculated from [17] using Equation 3.73

4-D
-t

R} = AC Co (3-74)

where

R, = rate of desorption {g/s)

A_ = area of desorption (cm?)

C, = initial concentration of the volatile component (g/cm?) in the polymer
D = diffusion coefficient (cm?/s)

t = time of exposure (s} of the polymer to the surrounding atmosphere

3.8.1 Permeability

Plastics are permeable by gases, vapors and Liquids to a certain extent. The diffusional
characteristics of polymers can be described in terms of a quantity known as per-
meability.
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The mass of the fluid permeating through the polymer at equilibrium conditions is given
by [7]

m=P'r'A'(P1_P2)

(3.75)
5
where
m = mass of the fiuid permeating (g)
p = permeability {g/(m : s - Pa}]
t = time of diffusion (s)
A = area of the film or membrane (m®)
p1»p, = partial pressures on the side 1 and 2 of the film (Pa)
s = thickness of the film (mm)

In addition to its dependence on temperature, the permeability is influenced by the
difference in partial pressures of the finid and thickness of the film. Other factors
influencing permeability are the structure of the polymer film, such as crystallinity, and
the type of fluid.

3.8.2 Absorption and Desorption

The process by which the fluid is absorbed or desorbed by a plastic material is time-
dependent, governed by its solubility and by the diffusion coefficient [7]. The period
until equilibrium value is reached can be very long. Its magnitude can be estimated by
the half-life of the process given by [7]

0049195

fos = D (3.76)
where

tg 5 = half life of the process
s = thickness of the polymer assumed to be penetrated by one side
D = diffusion coefficient

The value of t, ; for moisture in polymethyl methacrylate (PMMA) for
D=03-10"m%s and s=3mm

is 17.1 days when the sheet is wetted from one side only (7). However, the equilibrium
absorption takes much longer, as the absorption rate decreases with saturation.
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4 Designing Plastics Parts

The deformational behavior of polymeric materials depends mainly on the type and
magnitude of loading, time of application of the load, and temperature. The deformation
is related to these factors in a complex manner, so that the mathematical treatment of
deformation requires a great computational effort [1]. However, in recent times computa-
tional procedures for designing plastic parts have been developed using stress-strain data,
which were carefully measured by employing computer-aided testing of polymers [2]).

4.1 Strength of Polymers

The basic equation for calculating the design stress of a part under load is given by {1]

K
G, SOu=——r (4.1)
max $-A
where
K = material strength as a mechanical property
o, . = maximum stress occurring in the part
0, = allowable stress
S = factor of safety
A = material reduction factor

The relation between allowable stress and the polymer-dependent reduction factors can
be written as [1]

O = (4.2}
The factor Ag considers the influence of temperature on the strength of the material and

can be calculated from [1]

_ 1
A"‘l-{k(@-zon

where © = temperature, The limits of applicability of Equation 4.3 are 20 < © <100 °C.

(4.3)

The value k based on the reference temperature of 20 °C is given for the following materials
as [1]

PA6S =0.0112
PA6 =0.0125
PBT = 0.0095
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GR-PA and GR-PBT = 0.0071
POM = 0.0082
ABS =0.0117

The other reduction factors in Equation 4.2 consider the following effects:

The factor A, represents the effect of the time of static loading and can have following
values depending on time [1]:

time hours weeks months years

A 1.3 1.6 17 2

k]

The factor Ay, takes the effect of dynamic loading into account and lies in the range of
13 to 1.6.

The factor A, considers the influence of aging and has to be determined experimentally.

The reduction of strength caused by the absorption of moisture by the plastic can be
estimated from the factor Ayy. For unreinforced polyamides Ay, is roughly [1]

A, = 1 (4.4)

1-0.22f
with franging from 0 < f < 3 where f = weight percentage of moisture. The value of A, is
3.4 when fis greater than 3,

4.2 Part Failure

Usually stresses resulting from loading of the part are multiaxial. Because measured
material properties do not exist for combined stresses, the multiaxial state has to be
reduced to an uniaxial state by applying the principle of equivalence. According to Husgr,
von Mises and HEwky [1] the governing equation for the equivalent stress is

0,

1
HMH E[(O’l - 6,0 +(03 -0 + (0, ~ ;)1 (4.5)

where 0), 0, and 0; are normal stresses. The equivalent strain Z is defined by [3]

2
3
Materials, whose compressive stress is higher than the tensile stress, can be better described

by the conical or parabolic criterion [1]. The equivalent stress &, , according to the
conical criterion, is given as {1]

g

[(& - & P+ (& '53)2 +(5 - & ) (4.6)
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1+m

o, =1 [(m—l) (0,+0,+0;)t Alo, -0,  +(0, -0, +(0, —0'3)2}"'5] (4.7)
2m V2

"on
The parabolic failure criterion is defined by

m -

1
T = 5 (6, +03+0;) (4.8)

i{(m ) (0, +0,+0,) + ﬁ I:(o‘l —6,) +(0,-0,) +(0, —0'3]2]}

4m?

where m is the ratio of compressive stress to tensile stress.

Example

Figure 4.1 {1] shows a press fit assembly consisting of a metal shaft and a hub made of
POM. The joint strength can be determined as follows:

For the numerical values r,/r, = 1.6 and p = 22 N/mm?’ the equivalent stress is to be
calculated.

The tangential stress g, is given by

|:(1r'a,hfi)2 +I]
- (r, /5 —1

L

Substituing the values above
(L6" +1)
1.6% —1

The radial compressive stress g, is

G, =122- =50.2 (4.9)

0, = —p =22 N/mm?
&

4+ NN

/ Gt
Figure 4.1  Stress analysis in a press fit hub
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Substituting g, = 50.2 N/mm?>, 0, =~22 N/mm® and o = 0 in Equation 4.5 gives
_ 2
Toamm = N

The equivalent stress o,
. T
obtained from

[(50.2 +22)° +50.2% + 222] = 64.1 N/mm*

,according to Equation 4.7 for the conical failure criterion, is

_ 1
Toion =35

[0.4 (50.2-22)+% (5024227 +50.22 +222] = 58.98 N/mm*
with m = 1.4 for POM.

The yield point of POM is around 58 N/mm®. Thus, the assumed joint strength is too
high. In the case of deformation of the part by shear, the shear stress is given by [1]

=050 (4.10)

43 Time-Dependent Deformational Behavior

4.3.1 Short-Term Stress-Strain Behavior

As mentioned in Section 1.4, polymers are viscoelastic materials and their deformational
behavior is nonlinear. A typical stress-strain curve of POM under short-term loading is
shown in Figure 4.2. Curves of this type can be expressed by a fifth degree polynomial of
the form [4]

0 = PK, + PK, ||+ PK, -lef + PK, -|ef + PK4a-|ef* + PK, -|ef (4.11)

where PK,, ... PK; are polynomial coefficients dependent on the resin at a given tempera-
ture.

60
H/mmd

50 /’
. /
NV
N4

0 18

0 05 10 15 20 Y, z5
Strain £ —e=

123

Siress 6 —e

Figure 4.2  Stress-strain diagram for POM
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Stress & —=

Slrain £ —=
Figure4.3 Secant modulus [4]

The secant modulus (Figure 4.3) is given by
g =2 (4.12)

Setting PK;, = 0 it follows from Equation 4.11

E, = PK, + PK, ||+ PK,|ef + PK, -|ef +PK; | (4.13)

43.2 Long-Term Stress-Strain Behavior

The dimensioning of load bearing plastic components requires knowledge of stress-strain
data obtained under long-term loading at different temperatures. Retardation experiments
provide data on time-dependent strain in the form of creep plots, see Figure 4.4a. In
Figure 4.4b the stress is given as a function of time for different values of strain.
Isochronous stress-strain-time curves are illustrated in Figure 4.4c.

The critical strain method according to Mences [10] provides a useful criterion for
designing plastic parts. The experiments of MenGes and TaproGee [10] show that safe
design is possible when the critical strain is taken as the allowable deformation. The
corresponding tensile stress can be obtained from the isochronous stress-strain diagram.

The expression for calculating the time-dependent strain according to FINDLEY [8] is given
as

£(t)=AsinhBo+C-t"sinhDeo (4.14)

The power function of FINDLEY [2] is written as

et) = 29 4 i) 1@ (4.15)
EO

The function for the coefficient # is a fifth degree polynomial and that of n is a straight
line. With the Findley power function the long-term behavior of plastics up to 10° hours
can be described on the basis of measurements covering shorter periods of approx., 10°
hours [2].
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1 6] - 62 - 53
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- Ez___ p . h=heh
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- _’L__y)\

ih

Time t —e=

Shross & —=

h

Slress 6 —=

Time # —=
Figure4.4 Long-term stress-strain behavior [7]

Example [9]

Strain £ —=-

The minimum depth of the simple beam made of SAN shown in Figure 4.5 is to be
determined for the following conditions:

The beam should support a mid-span load of 11.13 N for 5 years without fracture and
without causing a deflection exceeding 2.54 mm.

b=12]

-

- o

41';’:11,13"
N - ./l d

- -

-

’/
" —|/
e
- -
/ ),

L7

Figure 4.5 Beam under midspan load {9]

Solution

The maximum stress is given by

_15-P-L
O-max— b-dz

(4.16)
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where

P = load {N)
L, b,d = dimensions in {mm) as shown in Figure 4.5

The creep modulus E_ is calculated from
p-L (
=— 4.17)
C 4-fb-
where fis deflection in mm, The maximum stress from Figure 4.6 after a period of 5 years
(= 43800 h}) is
Crax 23.44 N/mm’

60

N
[T ey
40

30 1--\

Fil]

10

Iniligl applied siress —

0

T T i L A
Time af rupture —=

Figure4.6 Creep curve for SAN [9]
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Time ¢ —e
Figure 4.7 Creep modulus for SAN (9]
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Working stress ¢, with an assumed safety factor S = 0.5:
o, =23.44- 0.5 = 11.72 N/mm’

Creep modulus E_ at ¢ < 0,, and a period of 5 years from Figure 4.7
E. = 2413 N/mm?

Creep modulus E_ with a safety factor § = 0.75:
E,. = 2413+ 0.75 = 1809.75 N/mm’

The depth of the beam results from Equation 4.16

15-2- LY (15-11.13-762\
go|L5-P-L z(—] 292 mm
bo 127 -11.72

The deflection is calculated from Equation 4.17

3 2
fe P-L - 11.13-76.2 - =215mm
4-E -b-d& 4-1809.75-12.7-2.92
fis smaller than 2.54 mm.
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5 Formulas for Designing Extrusion
and Injection Molding Equipment

5.1 Extrusion Dies

The design of extrusion dies is based on the principles of rheology, thermodynamics,
and heat transfer, which have been dealt with in Chapters 2 to 4. The strength of the
material is the determining factor in the mechanical design of dies. The major quantities
to be calculated are pressure, shear rate, and residence time as functions of the flow path
of the melt in the die. The pressure drop is required to predict the performance of the
screw. Information on shear rates in the die is important to determine whether the melt
flows within the range of permissible shear rates, Overheating of the melt can be avoided
when the residence time of the melt in the die is known, which also provides an indication
of the uniformity of the melt flow.

5.1.1  Calculation of Pressure Drop

The relation between volume flow rate and pressure drop of the melt in a die can be
expressed in the general form as (2]

Q=KG" &p" (5.1)
where
Q = volume flow rate
G = die constant
Ap = pressure difference
K = factor of proportionality in Equation 1.26
n = power law exponent Equation 1.30

It follows from Equation 5.1

Ap = (5.2)

5.1.1.1 Effect of Die Geometry on Pressure Drop

The die constant G depends on the geometry of the die. The most common geometries
are circle, slit and annulus cross-sections. G for these shapes is given by the following
relationships [2].
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i 3

1 R;+l
e
G. . =120 5.3
circle (4) 2L ( )
where
R =Radius
L = Length of flow channel
1 24
Wi H*"
Gy =|—["- 5.4
slit ( 6 ) 2L >4)

for w =20
H

where H is the height of the slit and W is the width.
For -g— < 20, Gy has to be multiplied by the correction factor F, given in Figure 5.1

The factor F, can be expressed as

F, =1.008 - 0.7474 - (%) +0.1638 (%T (5.5)
The die constant G, ;.. is calculated from

H=R,-R (5.6)
and

W=r{R;+R) (5.7)

where R, is the outer radius and R; is the inner radius. G then follows from

Equation 5.4

annulus

10 T
1 08 1 1
- H
5 06 I, S
s 04
g 02
0 04 08 12 16 20

Channel depth to width ralio H/W —=

Figure 5.1  Correction factor £, as a function of H/W[12]
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1 1 z,
G = (£ BB Ry =R 59
annulus 6 2L .

for values of the ratio #(Ry+ R,) / (Ry—- R;) 237

For smaller values of this ratio, G, has to be multiplied by the factor F, given in
Figure 5.1. The height H and width W are obtained in this case from Equation 5.6 and
Equation 5.7.

5.1.1.2 Shear Rate in Die Channels
The shear rate for the channels treated above can be computed from [3]

4Q

YVageae = TR (5.9
i 6-Q
Vo = 3 12 (5.10)
. 6-Q
Ya = (5.11)
annulus E(R0+Rl)(Rﬁ_Ri)2
The shear rate for an equilateral triangle is given by [4]
10 Q
Ywange ~ 3 g3 (5.12)
where d is the half length of a side of the triangle.
The relation for a quadratic cross-section is {4]
) 3 Q
=—— 5.13
Vonuwe =042 7 G13)

where a is the length of a side of the square,

In the case of channels with varying cross-sections along the die length, for example,
convergent or divergent channels, the channel is divided into a number of sufficiently
small increments and the average dimensions of each increment are used in the equations
given above [3].

5.1.1.3 General Relation for Pressure Drop in Any Given Channel Geometry

By means of the substitute radius defined by SchenkeL [5] the pressure drop in cross-
sections other than the ones treated in the preceding sections can be calculated. The
substitute radius is expressed by [5]
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1
n+l _ArH-Z n+3
4

Ry = T (5.14)

where

R, = substitute radius
A = cross-sectional area
B = circumference

5.1.1.4 Examples

The geometrical forms of the dies used in the following examples are illustrated in
Figure 5.2.

Example 1

It is required to calculate the pressure drop Ap of a LDPE melt at 200 °C flowing through
a round channel, 100 mm long and 25 mm diameter, at a mass flow rate of rir =10 g/s .

The constants of viscosity in the Equation 1.36 for the given LDPE are

Ay = 4.2968

A =-34709-107"
A, =—1.1008 - 107
A, = 1.4812-107
Ay =-11150-107

L

24
Figure 5.2 Flow channel shapes in extrusion dies
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129 -10°

48 10K

b,
melt density p, = 0.7 g/cm’

Solution

Volume flow rate O = - = 2% _ 1429 cm®/s = 1.429 - 10 m¥/s
p. 07

Shear rate 7, from Equation 5.9:

= -5
4-Q 41429107 o

AR Sl (0.0125)°
Shift factor a; from Equation 1.34:
ap = b, - exp(b, /T) = 1.29 - 10° exp[4860 /(200 + 273)} = 0.374

By Equation 1.37, the power law exponent # is

1 . . .
" =1+ A +2 A, lglar - 7,)+3 - Ay [lglar - ;,Va)]2 +4 A, [lgla; - }3)]3

Substituting the constants A, ... A, results in
n=1.832
Viscosity 77, from Equation 1.36

I, =lga; + Ag + A lglar - 7,) + 4, lglar - 7,)F
With ap = 0.374, 7, = 9.316 and the constants 4, ... A, we get

lg 1 =3.6514
17, =—4481Pa-s

Shear stress T from Equation 1.22
T=1, ¥, =4481-9.316 = 41745 N/m*
Factor of proportionality K from Equation 1.26:

K=2=3192-10"
T

Die constant G ;. from Equation 5.3:

!

RE+l 853 (0.0125 T
n .
Gcirc[e = (%)ﬂ : 2L = [%)1‘332 % =4,19- 10_5
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Pressure drop from Equation 5.2:

1

I . 1075)1832
ap=— L - (1429:107) = 668447 Pa = 6.68 bar

K" G (3.194 - 10781832 . 4,19 . 1075

circle

Example 2
Melt flow through a slit of width W= 75 mm, height H =1 mm, and length L = 100 mm.

The resin is LDPE with the same viscosity constants as in Example 1. The mass flow rate
and the melt temperature have the same values, 1 = 10 g/s and T = 200 °C. The pressure
drop Ap is to be calculated.

Solution

Tt 0.01

Volume flow rate Q = 200 =1.429-107 m*ss

m

Shear rate from Equation 5.10:

6-Q 6-1.429-107°

AW oors-oor P2 <
Shift factor ay from Equation 1.34:
ar=0.374
power law exponent # from Equation 1.37:
n=13.334
Viscosity 7, from Equation 1.36:
M,=257.6Pa-s

Shear stress T from Equation 1.22:
7= 294524.9 N/m’

Proportionality factor K from Equation 1.26:
K=6.669-107'

Correction factor F,

w =75
H
As the ratio W/H is greater than 20, the die constant which can be calculated from

Equation 5.4 need not be corrected.
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Gy = 2.1295 - 107
and finally the pressure drop Ap from Equation 5.2
Ap = 588.8 bar

Example 3

Melt flow through a slit with width W = 25 mm, height H = 5 mm, and length L = 100 mim.
The resin is LDPE with the same viscosity constants as in Example 1. The mass flow rate
m = 10 g/s and the melt temperature T = 200 °C. The pressure drop Ap is to be calculated.

Solution

Volume flow rate Q = i o0 =1.429 107 m’/s

P 700
Shear rate 7, from Equation 5.10:
7, =137.2s71
Shift factor a; from Equation 1.34:
ap = 0.374
Power law exponent s from Equation 1.37:
n = 2.604
Viscosity 7, from Equation 1.36:
1,=1044.7 Pa-s
Shear stress 7 from Equation 1.22:
T=143318.9 N/m’
Proportionality factor K from Equation 1.26:
K=5132-1072
Correction factor Fy;:

As the ratio W/H = 5, which is less than 20, the die constant G, has to be corrected.
F, from Equation 5.5:

F,=1.008 - 0.7474 - 0.2 + 0.1638 - 0.2” = 0.865

Gy = 5.208 - 107

Giit corrected = 5-208 - 107 - F, = 4505 - 10~
Pressure drop Ap from Equation 5.2:

Ap = 66.24 bar
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Example 4

Melt flow through an annulus with an outside radius R, = 40 mm, an inside radius
R; = 39 mm, and of length L = 100 mm.

The resin is LDPE with the same viscosity constants as in Example 1. The process
parameters, mass flow rate, and melt temperature remain the same.

Solution
Volume flow rate Q = m 00 1.429 -107° m°’/s
P 700
Shear rate ¥, from Equation 5.11;
- ]
. 6Q _ 6-1.429-10 = 34547 5~}

B Z(Ry+ R)(R, - R,-)z B 7 (0.04 + 0.039) 0.001

Shift factor 4 from Equation 1.34:
ar = 0.374

Power law exponent n from Equation 1.37:
n=2507

Viscosity 1, from Equation 1.36:
n,=579.43Pa-s

Shear stress 7 from Equation 1.22:
7=200173.6 N/m’

Factor of proportionality K from Equation 1.26:
K=13410-107"

Correction factor F,

7 (R, + R)

As the ratio

= 248.19 is greater than 37, no correction is necessary.
i

G

annulus

from Equation 5.8:

1 1 (R, - R )Eﬂ
F 4 - - Iy "
Gannulus = (E]ﬂ ) (Ro + Ri)" ’ 5L —

! 2
2907 1 o tl
(%}”‘” . (0.04 + 0.039)7307 - 0.0012%07

2-0.1
=1.443-107°
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Pressure drop Ap from Equation 5.2
Ap = 400.26 bar

Example 5

Melt flow through a quadratic cross section with the length of a side 2 = 2.62 mm. The
channel length L = 50 mm. The resin is LDPE with the following constants for the power
law relation in Equation 1.32:

1M, = Kog exp(=5-T)- A

B =0.00863

ng = 0.3286
Following conditions are given:
mass flow rate m =0.01g/s
melt temperature T =200°C
melt density Pm= 0.7 glcm®

The pressure drop Ap is to be calculated.

Solution
Three methods of calculation will be presented to find the pressure drop in this example.

Method a

With this method, the melt viscosity is calcilated according to the power law. Other than
that, the method of calculation is the same as in the foregoing examples.

Volume flow rate @ = 001 0.014286 cm’/s
Pm 07
) 6Q
Shear rate y, = Wi

For a square with W = H the shear rate #, is
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Viscosity i, from Equation 1.32:

7, = Ky exp(-f-T)- ™ -

= 135990 - exp(—0.00863 - 200) - 4.766"2%¢~1

= 24205.54 - 4.706°728¢1
=8484Pa-s
Shear stress T from Equation 1.22:
7= 40434.88 N/m’
Proportionality factor K from Equation 1.26:
K=4568-10"
Correction factor F,

As —g =1 is less than 20, the correction factor is obtained from Equation 5.5

F,=1.008-0.7474-1 + 0.1638 - 17 =0.4244
Die constant G, from Equation 5.4:

Gy =4.22- 107

Giit corrected = 04244 - 4.22:107° = 1.791 - 107
Pressure drop Ap from Equation 5.2

Ap = 35.7 bar

Method b
The shear rate #, is calculated from Equation 5.13

042 &  042-.262°
Viscosity 7, from Equation 1.32:
1, = 24205.54 - 5.674>*#51 = 754664 Pa - s
Shear stress T from Equation 1.22:

7=42819.6 N/m*
Power law exponent n from Equation 1.28:

n= —L = 3,043
Hp
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Proportionality factor K from Equation 1,26:
K=4.568-10"
The pressure drop Ap is found from
1

2 3 " Q°
A = =1"- 2L 5.15
Psquane ! [0.42) £ (5.15)
n an
with the die constant Gy,
1 3n
1{0.42Yn an
G =-=I"- 5.16
square 2( 3) 2L (5.16)
10042505 (262 Va5 1
42 Y300 62 Y3.043 -5
G === === -——-=1956-10
square 2( 3 J (1000] 0.05

In Equation 5.15

1 1
K" =(4.568 -10714)3043 = 4134 -107°

1 1
07 =(0.014286 - 107°)3083 = 2.643-10

Fnally Ap, .. from Equation 5.2:

1
O 2.643-107
Apsquare 1 4.134-107° .1.956-107°
K".G

The above relationship for shear rate developed by RamsteEmir [4] leads to almost the
same result as obtained by Method a.
Method ¢

In this method, a substitute radius is calculated from Equation 5.14; the pressure drop is
then calculated using the same procedure as in the case of a round channel (Example 1):

Substitute radins R,

1
(2”+l) n+2 n+3
At

= bid
o = =
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A =a* =262 = 6.864 mm’
B =4a4=42.62=10.48 mm

n =3.043
R, = 1363 mm

Shear rate ¥, from Equation 5.13:

- 40014286 _ ., o0 1
70,1363
Viscosity n, from Equation 1.32:
1, = 24205.54 - 7.183%°%'%"! = 6441.56 Pa - 5
Shear stress T from Equation 1.22:
7= 46269.73 N/m®

Factor of proportionality from Equation 1.26:

K=4568-107"
G iele from Equation 5.3:
1 3 a
( z ]3.043 . ( 1.363 )3.043
4 1000 -5
Giee = YT =1.885-10

Pressure drop from Equation 5.2:

2.643-107°
= = 33.92 bar
BPsquace 4.134-107° - 1.885-10°

This result shows that the relationship, Equation 5.14 [5], is sufficiently accurate for
practical purposes. This equation is particularly useful for dimensioning channels, whose
geometry differs from the common shape, that is, circle, slit or annulus. The procedure
of calculation for an equilateral triangle is shown in the following example:

Example 6

Melt flow through an equilateral triangular channel of length L = 50 mm. The side of the
triangle 2 d = 4.06 mm. Other conditions remain the same as in Example 5.

Solution

Substitute radius R;, from Equation 5.14 with
A =34 =7.1376 mm’
B=3-2d=1218mm
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n=3.043
Ry = 1.274 mm

Shear rate 7, from Equation 5.9:
_ 4-0014286
*omeoa274’

Viscosity 7, from Equation 1.32:

7, = 24205.54 - 8.8 = 5620.68 Pa - 5
Shear stress 7 from Equation 1.22:

7= 49462 N/m’
Factor of proportionality from Equation 1.26:

=88s"

K= 4568107
G e from Equation 5.3
1 3
[7:_}3.03 [1 274J3m3
Gy =22 1000 - 1.648 107
2-0.05

Pressure drop Ap from Equation 5.2:
i
Qr 2.643-107°

- — = 38.79 bar
4.134-107° - 1.648 - 10

Ap=—
K"-G

circle

Using the relation developed by RamsTENER [4] on the basis of rheological measurement
on triangular channels, Example 6 is calculated as follows for the purpose of comparing
both methods:

Shear rate from Equation 5.12:

Viscosity 1, from Equation 1.32:
7, = 24205.54 - 5.692°°2%1 = 7530.61 Pa - s
Shear stress 7 from Equation 1.22:
7= 42864.2 N/m’
Factor of proportionality from Equation 1.26:
K=4568-10""*
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Die constant Gyunget

3V P
1  dn
G. . =L (3], 5.17
tiangle = f3 (10} 2L (317)
—-3—+l
. (2.03)3‘043
1000

1 3 )3.043 -5
G =— .21 LTS - 1.75-10
wengle /3 (10] 2-0.05

Pressure drop Ap from Equation 5.2:

On 2.643- 10

1  4.134-107°-1.75-107
Km- Gtriangle

= 36.5 bar

This result differs little from the one obtained by using the concept of substitute radius.
Therefore this concept of ScHenkeL [5] is suitable for use in practice.

5.1.1.5 Temperature Rise and Residence Time
The adiabatic temperature increase of the melt can be calculated from
aT=— % (g (5.18)
10- oy, - Cpm
where

AT = temperature rise (K)

Ap = pressure difference (bar)

P, = melt density (g/cm®)

¢, = specific heat of the melt kJ/(kg - K)

pm
The residence time t of the melt in the die of length L can be expressed as
=L (5.19)
u

u = average velocity of the melt
Equation 5.19 for a tube can be written as
_4-L

T

R =tube radius

¥, = shear rate according to Equation 5.9

T (5.20)
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The relation of a shit is

T= ,6 L (5.21)
% H
H = height of slit
¥, = shear rate according to Equation 5.10

5.1.1.6 Adapting Die Design to Avoid Melt Fracture

Melt fracture can be defined as an instability of the melt flow leading to surface or volume
distortions of the extrudate. Surface distortions [34] are usually created from instabilities
near the die exit, while volume distortions {34] originate from the vortex formation at
the die entrance. Melt fracture caused by these phenomena limits the production of articles
manufactured by extrusion processes. The use of processing additives to increase the
output has been dealt with in a number of publications given in {35]. However, processing
aids are not desirable in applications such as pelletizing and blow molding. Therefore,
the effect of die geometry on the onset of melt fracture was examined.

The onset of melt fracture with increasing die pressure is shown for LDPE and HDPE in
Figure 5.3 [38]. As can be seen, the distortions appear differently depending on the resin.
The volume flow rate is plotted in Figure 5.4 [39] first as a function of wall shear stress
and then as a function of pressure drop in the capillary for LDPE and HDPE. The sudden
increase in slope is evident for LDPE only when the flow rate is plotted against pressure,
whereas in the case of HDPE it is the opposite. In addition, for HDPE the occurrence of
melt fracture depends on the ratio of length L to diameter D of the capillary. The effect
of temperature on the onset of melt fracture is shown in Figure 5.5 [36]. With increasing
temperature the onset of instability shifts to higher shear rates. This behavior is used in
practice to increase the output. However, exceeding the optimum processing temperature
can lead to a decrease in the quality of the product in many processing operations. From
these considerations it can be seen that designing a die by taking the resin behavior into
account is the easiest method to obtain quality products at high throughputs.

Design procedure

Using the formulas given in this book and in reference [33], the following design procedure
has been developed to suit the die dimensions to the melt flow properties of the resin to
be processed with the die.

STEP 1: Calculation of the shear rate in the die channel

STEP 2: Expressing the measured viscosity plots by a rheological model
STEP 3: Calculation of the power law exponent

STEP 4: Calculation of the shear viscosity at the shear rate in Step 1
STEP 5: Calculation of the wall shear stress

STEP 6: Calculation of the factor of proportionality

STEP 7: Calculation of die constant
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B —

1 "’“—/

5 LDPE
) =

HDPE

Figure 5.3  Irregularties of the extrudate observed at increasing extrusion pressure with LOPE and

HODPE [38]
InQ InQ
o o,
nQ nQ
= InApD; - I AD
LDPE HDPE

Figure 5.4 Volume rate vs. wall shear stress and vs. pressure drop in capillary for LOPE and HDPE [39]
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10t
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: /
% 10
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104 10° 10°
¢ Shear stress (N/m?)
Figure 5.5 Effect of temperature on the melt fracture {region 2) for HDPE

STEP 8: Calculation of pressure drop in the die channel and
STEP 9: Calculation of the residence time of the melt in the channel

Applications

Based on the design procedure outlined above, computer programs have been developed
for designing dies for various processes. The principles of the design methods are
illustrated below for each process by means of the results of the simulation performed
on the dies concerned. The designing principle consists basically of calculating the shear
rate, pressure drop, and residence time of the melt during its flow in the die and keeping
these values below the limits at which melt fracture can occur. This is achieved by changing
the die dimensions in the respective zones of the die, in which the calculated values may
exceed the limits set by the resin rheology.
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(bar) hear rat

200 t Residence time
180 —
1680 —
140 -
120 -
100 —
B0 —
60 —
40 ~
20 <

0.0005654 s

T T —T— 1T 7171 71— Die length (mm)
0 50 100 150 200 250 300 350 400 450 500

Flow rate: 350 kg/h LDPE

Temperature: 280 °C

Figure 5.6 Simulation results for a pelletizer die

a) Pelletizer Dies

The airm here is to design a die for a given throughput or to calculate the maximum
throughput possible without melt fracture for a given die. These targets can be achieved
by performing simulations on dies of different tube diameters, flow rates, and melt
temperatures. Figure 5.6 shows the results of one such simulation.

b) Blow Molding Dies

Figure 5.7 shows a blow molding parison and the surface distortion that occurs at a specific
shear rate depending on the resin. In order to obtain a smooth product surface, the die
contour has been changed in such a way that the shear rate lies in an appropriate range
(Figure 5.8). In addition, the redesigned die creates lower extrusion pressures, as can be
seen from Figure 5.8 [36].

¢) Blown Film Dies

Following the procedure cutlined above and using the relationships for the different
shapes of the die channels concerned, a blown film spider die was simulated (Figure 5.9).
On the basis of these results it can be determined whether these values exceed the boundary
conditions at which melt fracture occurs. By repeating the simulations, the die contour
can be changed to such an extent that shear rate, shear stress, and pressure drop are
within a range, in which melt fracture will not occur. Figure 5.10 and 5.11 show the
shear rate and the residence time of the melt along the flow path [37].

The results of simulation of a spiral die are presented in Figure 5.12 as an example. As in
the former case, the die gap and the geometry of the spiral channel can be optimized for
the resin used on the basis of shear rate and pressure drop.
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Figure 5.7 Surface distortion on a parison used in blow molding
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Figure 5.8 Die contour used for obtaining a smooth parison surface
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120
1.5
90
1 LDPE 2
& m =40 kgh
Tu =160 °C
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Figure 5.9 Calculated pressure drop in a spider die with different die gaps used for blown fitm
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Figure 5.10  Shear rate along spider die
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Figure 5.11 Residence time t of the melt as a function of the flow path |
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Figure 5.12  Results of simulation of a spiral die used for LLDPE blown film
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Manifold Manifold inlet radius: 24.84 mm
radius (mm) Pressure drop: 12.23 bar
Manifold angle: 60.34°
24
22
20
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16| e
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1411 o niell
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Length of the manifold {(mm})
Figure 5.13  Manifold radius as a function of the distance along the length of the manifold

d) Extrusion Coating Dies

Taking the resin behavior and the process conditions into account, the flat dies used in
extrusion coating can be designed following similar rules as outlined above. Figure 5.13
shows the manifold radius required to attain uniform melt flow out of the die exit as a
function of the manifold length [37].

5.1.1.7 Designing Screen Packs for Extruders

Screen packs are used in polymer processing extruders to remove undesired particulate
matter from the melt and are placed behind the breaker plate at the end of an extrusion
screw (Figure 5.16). Another important reason to implement screen packs is their
assistance in better back mixing of the melt in an extruder channel, which results from
the higher resistance offered by the screen to the melt flow. Better back mixing in turn
improves the melt homogenity. In addition, screen packs may also be used to attain higher
melt temperatures to enable better plastication of the resin. Owing to the intimate
relationship between melt pressure and extruder throughput it is important to be able to
predict the pressure drop in the screen packs as accurately as possible.

Design Procedure

The volume flow rate § through a hole for a square screen opening (Figure 5.14) is given
by [42]

§=(400-6.45- M)/(3.6 - p,, - m’ -7 - DY)
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Table5.1  Dimensions of Square Screens [41]

Mesh size Sieve opening Nominal wire diameter
ram mm
42 0.354 0.247
100 0.149 0.110
200 0.074 0.053
325 0.044 0.030
25.4 mm
(1 inch)

Figure 5.14 Mesh of a wire-gauze screen

my,= 200 100 100 42 42

meit

Figure 5.15  Screen pack with screens of varied mesh size

The shear rate of the melt flow for a square opening is calculated from

Fiquare = (310.42) - 4/(0.001- 43)

By means of these equations and the design procedure outlined in Section 4.1, following
examples were calculated and the results are shown in Figure 5.17 to Figure 5.20.
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Die _\R\ // ——— Flange
N
\\ ez
k \
. S
Breaker plate — ] N
SN\ 77
Screen pack Melt pressure
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Flgure 5.18  Posltion of screan pack in an extruder [43]
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Figure 5.17  Effect of polymer type on pressure drop Ap in the screen pack
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Figure 5,18  Extruder throughput
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Figure 5.19  Effect of the mesh size on the pressure drop in the screen
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Figure 5.20 Effect of reduced screen area on the pressure drop in the screen

5.2 Extrusion Screws

In this chapter, formulas for the quantities often required when dimensioning extrusion

screws are lllustrated by specific examples.

5.2.1 Solids Conveying

Under the assumptions that:

{(a) the polymer moves through the screw channel as a plug,
{b) there is no friction between the solid plastic and the screw, and

(c) there is no pressure rise,
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the maximum flow rate (Os Jmax (s€€ Figure 5.21 and Figure 5.22) can be calculated from
(6]

(Qs)max=772-H°Db(Db—H]-sin¢cos¢-—W—-N (5.22)

W+ wy

The actual flow rate Qs is given by [6]

tan© - tang w

2, =7 H Dy (D, - H)- - 5.23

R b (Do = H) tan® + tang W + wp 5-23)
The conveying efficiency 1 can be expressed as

e = =2 (5.24)

G

Figure 5.21  Screw zone of a single screw extruder [7]

Figure 5.22 Movement of solids in a screw channel after Tabwor [6)
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In practice, this efficiency is also defined as

G
hp=— (5.25)
NV p

where
G, = mass flow rate
N = screw speed
V, = volume of the screw channel
P = bulk density
Example
The geometry of the feed zone of a screw, Figure 5.22, is given by the following data [6)
barrel diameter Dy, =50.57 mm
screw lead s =50.57 mm
number of flights v =1
root diameter of the screw D, =34.92 mm
flight width Wpr = 5.057 mm
depth of the feed zone H = 7823mm

The maximum specific flow rate and the actual flow rate are to be calculated.

Solution

Helix angle ¢

S tan~!| =3 | = m'l(ﬂJ =17.66 °
D, z - 50.57

Width of the screw channel W:

_ scosg

W - wFLT

= 50.57 - cos 17.66 — 5,057
= 50.57 - 0.953 — 5.057 = 43.13 mm

Maximum specific flow rate from Equation 5.22:

L) 222078235057 (5.057 - 0.7823) - 03034 - 0,953 2212
N) o 4.819

= 43.19cm’ /1ipm
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Taking the bulk density p,, = 0.475 gfcm’ into account, the specific mass flow rate becomes

G

= =2052

[ N J g/rpm
max

The feed angle © is required to calculate the actual flow rate. With the assumptions already
made and assuming equal friction coefficients on screw f, and barrel f, the approximate
feed angle may be calculated from [6]

cos® =Ksin@ + sin¢(K + -g:—cot ¢) (5.26)

where
K= D(sing + f, cos¢)
Dy(cos¢ — f, sing)
With £, = f, = 0.25 and the average diameter D = Dy — H = 5.057 — 0.7823 = 4.275 cm

(5.27)

_ 4.275(0.3034 + 0.25-0.953) _ 0

= = 0.522
5.057 (0.953 — 0.25 - 0.3034)
With K =0.522 and D, _3492 0.6906 we obtain from Equation 5.26
Dy, 5057

cos© = K sin® + 0.816

0 ~15.8°
Inserting © = 15.8° into Equation 5.23

Qs _tan©-tang w

2
=g H-D -H
N b (Dy ) tan® + tangd W+ wyy
gives
tan15.8 - tan17.66  4.313
tanl5.8 + tanl17.66 4.819

%. =7 - 0.7823 - 5.057 (5.057 — 0.7823) -

= 22.4 cm*/rpm
The actual specific mass flow rate using the bulk density p_, = 0.475 g/cm’ is therefore

G
— =10.62 g/frpm
= 10628/
The conveying/ efficiency 7} is
e = 1062 _ 0.52

20.52
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5.2.2 Melt Conveying

Starting from the parallel plate model and correcting it by means of appropriate correction
factors [7], the throughput of melt in an extruder can be calculated. Although the following
equation for the output applies to an isothermal quasi-Newtonian fluid, it was found to
be useful for many practical applications [3].

For a given geometry of the melt zone {Figure 5.21), the output of a melt extruder or
that of a melt pumping zone of a plasticating extruder can be determined as follows
[3,7]

Helix angle ¢:

- tan |
@ = tan [ﬂDb] {5.28)

Volume flow rate of pressure flow Qp {m>/s):

-7-D, H [1-QJ-sin2¢-Ap-1o-“
5

), = 5.29

Q, 2oL (5.29)
Mass flow rate 1, (kg/h):

rit, = 3600 - 1000 - Q, - Py, (5.30)
Drag flow Qd (m’/s):

72D, -N-H? [1—QJ-sin¢-cos¢-10‘9
: s
= 5.31

Qq 260 (5.31)
Mass flow rate 1, (kg/h):

tiy = 36001000 Q; - p., (5.32)
The leakage flow through the screw clearance is found from the ratios

ay = & (5.33)

Q4
and
é
=— 5.34
T=5 (5.34)

The extruder output # is finally calculated from

hh=6-10" 7° - D} -N-H(l—%)-pm .singcosg-(1—a—J)/2 (5.35)
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The shear rate required for determining the viscosity ), at the given melt temperature T
is obtained from

. - w- Db - N

o=t (5.36)
Symbols and units used in the formulas above:
Dy: Barrel diameter mm
H: Channel depth mm
e Flight width mm
s Screw lead mm
& Flight clearance mm
L Length of melt zone mm
v: Number of flights -
Ap: Pressure difference across the melt zone bar
Yai Shear rate !
Q,»Qy:  Volume flow rate of pressure flow and drag flow, respectively m’/s
fity, i3 Mass flow rate of pressure and drag flow, respectively kg/s
m: Extruder output kg/h
n,: Melt viscosity Pa-s
ay: Ratio of pressure flow to drag flow -
T: Melt temperature °C
N: Screw speed min~
Example

For the following conditions the extruder output is to be determined:
Resin: LDPE with the same constants of viscosity as in Example 1 in Section 4.1.1.
Process parameters:

Screw speed N =80 min™" (rpm)
Melt temperature T =200°C
Melt pressure Ap = 300 bar

Geometry of the metering zone:

Dy, =60 mm; H=3 mm; e =6 mm; s = 60 mm; 8y =0.1 mm; L =600 mm; v=1

Solution

7, =838s" Equation 5.36

ar =0.374 Equation 1.34

7, =1406.34Pa-s Equation 1.36

¢ =17.66° Equation 5.28

, =-3.146kg/h Equation 5.29 and Equation 5.30
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my =46.42kg/h Equation 5.31 and Equation 5.32
m  =418kgh Equation 5.33, Equation 5.34 and Equation 5.35

Leakage flow rit, = ritg + rir, = rir = 1.474 kg/h

5.2.2.1 Correction Factors

To correct the infinite parallel plate model for the flight edge effects, following factors are
to be used along with the equations given above:

the shape factor for the drag flow F; can be obtained from [8)} with sufficient accuracy

H

Fy =1-0.571 W {5.37)
and the factor for the pressure flow F,
H
Fp =1-0.625 W (5.38)

The expressions for the corrected drag flow and pressure flow would be

Qdk =Pd'Qd
and
Qi =5 Qp

The correction factor for the screw power, which is treated in the next section, can be
determined from [9]

F,=exp(x)—x +22x-1.05x (5.39)
with
H
x=—
w

Equation 5.3% is valid in the range 0 < H/W < 2. For the range of commonly occurring
H/W-ratios in extruder screws, the flight edge effect accounts for only less than 5% and
can therefore be neglected [8]. The influence of screw curvature is also small so that F,
can be taken as 1.

Although the above mentioned factors are valid only for Newtonian fluids, their use for
polymer melt flow is justified.

52.2.2 Screw Power

The screw power consists of the power dissipated as viscous heat in the channel and
flight clearance and the power required to raise the pressure of the melt. Therefore, the
total power Z,, for a melt filled zone [10] is
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Zy=Z +Zyr+ Zy (5.40)
where

Z, = power dissipated in the screw channel
ZFLT = power dissipated in the flight clearance
Zyp, = power required to raise the pressure of the melt

The power dissipated in the screw channel Z, is given by [10]

v.ml Dis N?> W .7 -AL(F, cos’ ¢ + 4sin’ ¢)

Ze= 36 10" - 8y - sing (5.41)
The power dissipated in the flight clearance can be calculated from {10]

Zpyy = L2 D N ey iy - AL (5.42)

36-10"° - dg - sing

The power required to raise the pressure of the melt Z,, can be written as

Z,, =100-Q, - Ap (5.43)
The flight diameter Dy is obtained from

Dpyr =Dy~ 2 8pr (5.44)
and the channel width W

W= %cosqb - Wpr {5.45)

The symbols and units used in the equations above are given in the following example:

Example

For the following conditions the screw power is to be determined:

Resin: LDPE with the constants of viscosity as in Example 1 of Section 5.1.1.4
Operating conditions:

screw speed N =80rpm
melt temperature T =200°C
die pressure Ap = 300 bar

Geometry of the melt zone or metering zone:

D =60 mm; H =3 mm; ¢ = 6 mm; 5 = 60 mm; Syr = 0.1 mm; AL = 600 mm; v=1
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Solution
Power Z_ in the screw channel:

Dgrr = 59.8 mm from Equation 5.44
Shear rate in the screw channel 7,:

¥ =83.8s™ from Equation 5.36

ar = 0.374 from Equation 1.34
Viscosity of the melt in the screw channel 7.

1. = 1406.34 Pa - s from Equation 1.36
Channel width W:

W = 51.46 mm from Equation 5.45
Number of flights v:

v=1
Length of the melt zone AL:

AL = 600 mm
Faktor F,:

F, =1 for H_3 _ 0,058 from Equation 5.39
W 5146

Helix angle ¢
¢ = 17.66° sing = 0.303 from Equation 5.28

Power in the screw channel Z_ from Equation 5.41:

Z, =1-7°-59.8% - 80% - 51.46 - 1406.34 - 600 -

(1-cos?17.66° + 45in” 17.66°)

= 3.84 kW
Power in the flight clearance Zp
Flight width wg;y (Figure 5.22):
Werr = € €os ¢ = 6 - cos 17.66° = 5.7 mm
Shear rate in the flight clearance 4 :

7 Dy-N_7-60-80

= =2513.3s7"
Fux =g Sar  60-0.1

3610 - 3.5in17.66°
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Shift factor ay:
ap=0.374 at T = 200 °C from Equation 1.34
Viscosity in the flight clearance ng 1
Ngr = 219.7 Pa's from Equation 1.36
Length of the melt zone AL:
AL = 600 mm
Zgyr from Equation 5.42:
Zor = 1-7* -59.8 -302 (600572197 _\ o\
36-107 -0.1-0.303

Power to raise the melt pressure Zy,

Pressure flow Qp :
Qp from the Example in Section 4.2.2
Q, =1.249-107° m’/s
Die pressure Ap:
Ap= 300 bar
Z,, from Equation 5.43:
Zpp =100 - 1.249 - 107 - 300 = 0.0375 kW

Hence the power Z,, is negligible in comparison with the sum Z_ + Zg;y.

5.2.2,.3 Heat Transfer between the Melt and the Barrel

Te estimate the power required to heat the barrel or to calculate the heat lost from the
melt, the heat transfer coefficient of the melt at the barrel wall is needed. This can be
estimated from [11]

_ N_ V[, - (Ti=T) - exp(B)}
asz_ﬂm[60.x.a) [l (Tb—Tm) (5.46)
where the thermal diffusivity 2
= # 5.47
T, pa (547

and the parameter 8

—6 2
p= _19_24?4_.*_3_1:{ (5.48)
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Indices:

m: melt

£ melt ilm

b: barrel

Example with symbols and units

Thermal conductivity A, =0.174 W/(m - K}
Specific heat Com= 2 kl/(kg - K}
Melt density P = 0.7 g/cm’

Thermal diffusivity a from Equation 5.47:
a=1243-107 m¥s

Flight clearance &y = 0.1 mm
Screw speed N =80rpm

Parameter f§ from Equation 5.48:

B=0.027
For T;=137.74°C, T,, = 110 °C and T, = 150 °C
¢, from Equation 5.46:
2 \05
a, =0.174] 210 {1 _ (1377 -110)ft exp(—0.027)}]
60 - 1 -1.243 150 — 110

= 315.5 W/(m? - K)

5224 Melt Temperature

The exact calculation of melt or stock temperature can be done only on an iterative basis
as shown in the computer program given in [9]. The following relationships and the
numerical example illustrate the basis of calculating the stock temperature. The result
obtained can only be an estimate of the real value, as it lacks the accuracy obtained by
more exact iterative procedures.

Temperature rise AT:

3600(Z, + Zp + Ny)

AT =Ty =Ty = , (5.49)
fit - Com
Heat through the barrel or heat lost from the melt:
NH = & 'E'DFLT 'AL(Tb -TEIN) {5.50)

10% - cos ¢
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Example for calculating N, with symbeols and units
@, = 315.5 W/(m? - K); Dyyp = 59 mmy; AL = 600 mm; T, = 150 °C; tom = 2 KIf(kg - K)
Stock temperature at the inlet of the screw increment considered:
T, = 200°C
Ny from Equation 5.50:
_ 315.5 -7 - 59.8 - 600 - 50 _
10° - cos17.66°

AT with the values Z_=3.84 kW, Zg 1 = 1.56 kW and 7 = 41.8 kg/h from the earlier
example from Equation 5.50

Ny —1.86 kW (heat loss from the melt)

T= 3600 3.54

=152.4°C
41.8-2
Stock temperature at the outlet of the screw increment considered T:
T, = Ty + 1524°C

Melting point of the polymer Ty, = 110°C
Hence, T, =110 + 152.4 = 262.4 °C
Average stock temperature T:
T = L, + Ty _ 200 +262.4
2

As already mentioned, this result can only be an estimate because the effect of the change
of temperature on the viscosity can be calculated only through an iterative procedure as
shown in [9].

=2312°C

5.2.2.5 Melt Pressure

For a screw zone of constant depth the melt or stock pressure can generally be obtained
from the pressure flow by means of Equation 5.29. However, the following empirical
equation [10] has been found to give good results in practice:

A n
F-2-7,- Al _|QP|(2:;R+1)-(HR+H§)-109 ®

= 107 (5.51)
42 sing (Hy, + Srr) W (Hoy + &) -1y v (
where
[/}
Ty = (5.52)

The sign of Ap corresponds to that of the pressure flow Qp .
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Example with symbols and units
a) Screw zone of constant channel depth (metering zone)

Empirical factor F, =0.286
Melt viscosity in screw channel n, =1400Pa-s
Shear rate in channel y =845
Length of screw zone (or of an increment) Al =600 mm
Helix angle ¢ =17.66°
Channel depth at the outlet of the zone or increment  H_, =3 mm
Flight clearance Sy = 0.1 mm
Pressure flow Qp =1.249-10 6 m’/fs
Reciprocal of the power law exponent n ng =05
Ratio of channel depths at the outlet (H_,,,)
and inlet (H;,) of the zone or increment Hy Hy =1 {(constant depth)
Width of the channel W =5146 mm
Thickness of the melt film & =0
Number of flights v =1
1}, from Equation 5.52:

1= 10 2g3
Ap from Equation 5.51:

) 9 L5
Ap = 0.'286 2:600-12831| 1.249 - 10°5(2- 05+ 1) (1+1)210 1075
5in17.66%(3+40.1) 5]1.46-3%.0.5-1

=218 bar
b) Screw zone of varying depth (transition zone)
H, =9mm; H,, =3 mm; Al=240 mm; 7 = 1800 Pa- 5; = 425"
7, from Equation 5.52:

1800
ﬂp=zzo'—5":i“=ll665
Ap from Equation 5.51:
0.5
2 vl
pp={0286:2: 24011665 |, ) \g 106, (g)*{;} _ 102 10
sin17.66° (3 + 0.1) 9/ \9/ ] 5146-3*.05
= 37.2 bar

A more exact calculation of the melt pressure profile in an extruder should consider the
effect of the ratio of pressure flow to drag flow, the so called drossel quotient, as shown in
[10].
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5.2.3  Melting of Solids

Physical models describing the melting of solids in extruder channels were developed by
many workers, notably the work of TADMOR [6]. RAUWENDAAL summarizes the theories
underlying these models in his book [8]. Detailed computer programs for calculating
melting profiles based on these models have been given by Rao in his books [3, 9].

The purpose of the following section is to illustrate the calculation of the main parameters
of these models through numerical examples. The important steps for obtaining a melting
profile are treated in another section for a quasi Newtonian fluid.

5.2.3.1 Thickness of Melt Film

According to the Tadmor model {6] the maximum, thickness of the melt film (Figure 5.23)
is given by

0.5
2 4 (T, = T,) + 75 V) 107 W (55
0 Vg o[y T — T+ |
Example with symbols and units
Thermal conductivity of the melt Ay =0.174 W/(mK)
Barrel temperature T, =150°C
Melting point of the polymer T =110°C
Viscosity in the melt film T Pa-s
Shear rate in the film ¥ s
Velocity of the barrel surface Vi cm/s
Velocity components Vie Vi, cm/s (Figure 5.24)
Velocity of the solid bed V,, cm/s
Cutput of the extruder G = 16.67 gfs
Average film thickness o mm
Temperature of the melt in the film T; °C
Average film temperature T, °C
Depth of the feed zone H, =9mm
Width of the screw channel w = 51.46 mm
Melt density P =07 glm’
Density of the solid polymer P, = 0.92 g/cm’
Specific heat of the solid polymer Cps =22kJ/kg - K)
Temperature of the solid polymer T, =20°C
Heat of fusion of the polymer i, =125.5 kJ/kg
Maximum film thickness Onax €M
Indices:
m: melt

s: solid
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Figure 5.23 Temperature profile in the melt film after Taomon [6]
a:solid bed, b: barrel surface, ¢: melt film, d: solid melt interface
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Figure 5.24  Velocity and temperature profiles in the melt and solid bed after Taomor [6]
a: solid melt interface, b: cylinder, ¢ solid bed

Following conditions are given:
The resin is LDPE with the same constants of viscosity as in Example 1 of Section 4.1.1.4.
The barrel diameter Dy, is 60 mm and the screw speed is 80 rpm.

_Z-Dy-N _x-60-80
10- 60 1060

b = 25.13 cm/s



120 5 Fermulas for Designing Extrusion ... [Refs. on p. 161]

Vix = V4, sing = 25.1-5in17.66° = 7.62 cm/s
Vi = Vi, cos@ = 25.1 - cos17.66° = 23.95 cm/s

10-G _ 100-16.67

= = = 3.91 cm/s
W-H, -p, 51.46-9-0.92

Relative velocity V; (Figure 5.24):
V,=(Vy +V; -2V, -V, - cosp)*®
=(25.13% + 3.91* - 2-25.31- 3.91 - cos17.66°)>°
= 21.44 cm/s
Temperature T,:
7Tt Ty _150+110
: 2
_Y
%

Starting from an assumed film thickness of 0.1 mm and using the temperature resulting
when heat generation is neglected, the viscosity in the film is estimated first. By changing
the film thickness and repeating this calculation, the final viscosity is obtained [3].

=130°C

%

This iteration leads to

; _Yi_z-2144-10
3 0.299

7 =351.84Pa-s

=14345""

O, from Equation 5.53
Frnax = { [2 -0.174(150 — 110) + 351.84 - 21.447 - 10“‘]

0.5
_ 51.46
10° - 7.62-0.7[2.2(110 - 20) + 125.5]

= 0.0299 cm, or 0,299 mm

Temperature in Melt Film

Taking the viscous heat generation into account, the temperature in melt film can be
obtained from [6)
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~ = Wt.p-V?
=T + —— 1 5.54
f a 12’2‘“ ( )
-4 . 2
I-T-= 107 -351.84 -21.44° _ .o
12-0.174

T = (@) +7.75=1375°C

As seen from the equations above, the desired quantities have to be calculated on an
iterative basis. This is done by the computer program given in [3).

5.2.3.2 Melting Rate

The melting rate is described by TADMOR [6) by the parameter ¢, , which is expressed as

0.5
Vi Pin [ Am (B = T + 057 - V- 107 ]

. (5.55)
100 - 2[c,, (T, — T,) + iy}

# =

The numerator represents the heat supplied to the polymer by conduction through the
barrel and dissipation, whereas the denominator shows the enthalpy required to melt
the solid polymer. The melting rate increases with increasing ¢, .

By inserting the values given above into Equation 5.55 we obtain

0.5
_]7.62-0.7[0.174 (150 — 110) + 0.5 - 351.84 - 21.44” - 107*]
% 100 - 2[2.2 (110 — 20) + 125.5]

8
=0.035—"—
cm'
5.2.3.3 Dimensionless Melting Parameter
The dimensionless melting parameter ¥ is defined as [6]

¢P . Hl . WO.S
=2 (5.56)
101‘5 . G
with
9, = 0.035 g/(cm'”+ )
H, =9 mm
W =51.46 mm

G =16.67gfs
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we get
W= 0.004

The dimensionless parameter is the ratio between the amount of melted polymer per
unit down channel distance to the extruder output per unit channel feed depth.

5.2.3.4 Melting Profile

The melting profile provides the amount of unmelted polymer as a function of screw
length (Figure 5.25) and is the basis for calculating the stock temperature and pressure.
It thus shows whether the polymer at the end of the screw is fully melted. The plasticating
and mixing capacity of a screw can be improved by mixing devices. Knowledge of the
melting profile enables to find the suitable positioning of mixing and shearing devices in
the screw |21].

The following equation applies to a screw zone of constant depth [6]

2
Kow _ Xin [ _ Y &2 (5.57)
w o wl 2-H
and for a tapered channel [6]
2
Kow _ Xig K_(K_l] Hi, (5.58)
w wla (A H,,
where
a-H-H (5.59)
zZ

Melt Solid bed Melt Fim

! Borrel
{
n
2
N i
> 5 Screw

7
X
s 1

4

Maal distance along the screw —e-
o Cross-seclion of screw channel

Figure 5.25  Sclid bed or melting profiles X/W and G,/G [21]
G: total mass flow rate, G,. mass flow rate of solids
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Figure5.26 Three-zone screw (8]

The parameter ¥ is obtained from Equation 5.56.

Symbols and units;

Xowp Xin mm  Width of the solid bed at the outlet and inlet of a screw increment
respectively

w mm  Channel width

v Melting parameter

Az mm Downchannel distance of the increment

H,H,, mm Channel depth at the inlet and outlet of an increment

H;,H, mm Channel depth of a parallel zone (feed zone) and depth
at the end of a transition zone (Figure 5.26)

A Relative decrease of channel depth, Equation 5,59

z mm Downchannel length of a screw zone

a) Constant channel depth
For

H, =9 mm; X;)/W=1; Az =99 mm and ¥ = 0.004 from Section 4.2,3.3, X /W can be
calculated from Equation 5.57:

2
Xow 1(1_ 0.004—99] - 0.96
w 2-9

This means that at a distance of Az = 99 mm, 4% of the solids were melted.

b) Varying channel depth
For the values

H, =9mm

H, =3 mm

Zz = 1584 mm

Xin’ W = 0.96

H, =9 mm

H,, =8625
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¥ can be obtained from Equation 5.56:

v = ¢, -H -W™  (035.9.51.46"°
105 (X_m) G 107-096-16.67
w

= 0.00447

The relative decrease of the channel depth A is calculated from Equation 5.59:

H-H, 9-3
4 1584
and X, /W from Equation 5,58

2
Xout _ o.gg! 0:0047 _[ 0.0047 _ 1} 9
w 0.00379 10.00379 8.625

A= = 0.00379

Assuming a constant velocity of the solid bed, the mass flow ratio G/G results from

oy
>
]

A L (5.60)
G WH,
where
G; = mass flow rate of the solid polymer g/s
G = througput of the extruder g/s
X = average of X, and X, mm
H =average of H,,,,, and H, mm
For a zone of constant depth it follows that
G_X (5.61)
G W
a) Constant depth
X = (.‘I.S(ﬁ + h) =0.5-1.96 = 0.98
w w W
-Gi =098
G
b) Varying depth
X =05 [& + h] = 0.5:(0.96 - 0.953) = 0.9565
w w w
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H, 9
G _XH _ 9366
G WH,

The profiles of stock temperature and pressure can be calculated from the melting profile
by using the width of the melt-filled part of the channel in the equations given in
Section 5.2.2 [10].

5.24 Temperature Fluctuation of Melt

Temperature and pressure variations of the melt in an extruder serve as a measure for the
quality of the extrudate and provide information as to the performance of the screw.

The temperature variation AT may be estimated from the following empirical relation,
which was developed from the results of SQuRes’ experiments [12] conducted with 3-zone
SCrews:

U] R T 62
914.31 Ng —0.024

This relation is valid for 0.11 < N, < 0.5.

The parameter Ny, is given by
N, —147-10“‘2'§—Z~Ii (5.63)
Q - G H .
where

AT = temperature variation (°C)

D, = barrel diameter {cm)

G = extruder output (gfs)

L =length of screw zone in diameters
H = depth of the screw zone (cm)

Example
Following values are given:

Db=6cm
G =15gfs
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L depth cm LiH
9 0.9 10

3 0.6 (mean value) 3.33
9 0.3 30

Hence Z% =43.33

Ng, from Equation 5.63:

N, =147-10~ - 2% . 4333 = 0.153
Q 15

AT from Equation 5.62:
5 1
T=—[ 3 :|=7.22°C0ri3.6l°C
914.31-0.153° - 0.024

The constants in the Equation 5.62 and Equation 5.63 depend on the type of polymer
used, For screws other than 3-zone screws the geometry term in Equation 5.63 has to be
defined in such a way that Ny, correlates well with the measured temperature fluctuations.

5.2.5 Scale-up of Screw Extruders

Based on the laws of similarity, PEarson [13] developed a set of relationships to scale-up
a single screw extruder. These relations are useful for the practicing engineer to estimate
the size of a larger extruder from experimental data gathered on a smaller machine. The
scale-up assumes equal length to diameter ratios between the two extruders. The
important relations can be summarized as follows:

\(1-5)i(2—35)

H, _{D (5.64)
H D

—{2-25)/(2-335)
N, D (5.65)
N, D,
. \(3-553(2-35)
LA Y (5.66)
"y D,
H (1-$)H{2—3s)
s (D2 (5.67)
Hy D
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where

Hp = feed depth

H = metering depth
D = screw diameter
N =screw speed

Indices: 1 = screw of known geometry and 2 = screw to be determined.
The exponent s is given by
s=0.5(1—-ny)

where np is the reciprocal of the power law exponent n. The shear rate required to
determine #n is obtained from

? _ - Dl " Nl
°  60-H,
Example

Following conditions are given:

The resin is LDPE with the same constants of viscosity as in Example 1 of Section 5.1.1.4.
The stock temperature is 200 °C. The data pertaining to screw 1 are:

D, =% mm; Hp=12 mm; H; = 4mm

feed length =9D,
transition length =2 D,
metering length =9 D,
output 1y =130 kg/h
screw speed N; = 80 1pm

The diameter of screw 2 is D, = 120 mm. The geometry of screw 2 is to be determined.

Solution

The geometry is computed from the equations given above [3]. It follows that

D, = 120mm
Hp, = 1441 mm
H, = 48mm
i, = 192.5kg/h
N, = 555rpm

Other methods of scaling up have been treated by ScHENKEL [29], FENNER [30], FiscHER
[31], and Potente [32].

Examples for calculating the dimensions of extrusicn screws and dies are illustrated in
the following figures:
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5.2.6 Mechanical Design of Extrusion Screws
5.2.6.1 Torsion

The maximum shear stress 7,,,,,, which occurs at the circumference of the screwroot asa
result of the torque My, is given by [8]

e = 2 (5.68)
where R = root radius of the screw.
The maximum feed depth H, ., can be computed from [8]
1
H,, =05-D- (i ‘:_'f: Js (5.69)
where
D = diameter

T,u = allowable shear stress of the screw metal

Example [8]
The maximum feed depth is to be calculated for the following conditions:
D = 150 mm; My = 17810 Nm; 7,, = 100 MPa;

H_,., is found from Equation 5.69:

1
g -1 150 _[ 2-17810 )5
2 1000 \7x-100-10°

= 0.075 —0.0485 = 0.0265m or 26.5mm

5.2.6.2 Deflection
Lateral Deflection

The lateral deflection of the screw (Figure 5.34) caused by its own weight can be obtained
from [8]

2:1000-g-p- I

Y(L)=
(L) s

(5.70)
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Figure 5.34 Lateral deflection of the screw as cantilever [8]

Numerical example with symbols and units [8]

g =9.81m%s acceleration due to gravity

p =7850kg/m’ density of the screw metal

L =3m length of the screw

E =210-10"Pa elastic modulus of the screw metal
D =015m screw diameter

Inserting these values into Equation 5.70 we get

1
Y(L)=2 1000 9.891 785(2) 3 — 2.64 mm
210-10° - 0.15

This value exceeds the usual flight clearance, so that the melt between the screw and the

barrel takes on the role of supporting the screw to prevent contact between the screw and
the barrel [8].

Buckling Cuased by Die Pressure
The critical die pressure, which can cause buckling, can be calculated from [8]
_07°.7-E

Numerical example [8]

E =210-10°Pa elastic modulus of the screw metal
L/ID =35 length to diameter ratio of screw

Px (5.71)

Py from Equation 5.71:

107 - 7%.210-10°

PYREYS = 264.36 bar

As can be seen from Equation 5.71, the critical die pressure py decreases with increasing
ratio L/D. This means, that for the usual range of die pressures (200-600 bar) buckling
through die pressure is a possibility, if the ratio L/D exceeds 20 [8].
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Screw Vibration

When the screw speed corresponds to the natural frequency of lateral vibration of the
shaft, the resulting resonance leads to large amplitudes, which can cause screw deflection.
The critical screw speed according to [8] is given by

1
D(EY:
Ny = o.ssL—z(;} (5.72)

Substituting the values for steel, E = 210 - 10° Pa and p = 7850 kg/m’ we get
4549.5

@

For D = 150 mm and % = 30, Ny is found from Equation 5.73

Ny = (5.73)

45495
K 015302

This result shows that at the normal range of screw speeds vibrations caused by resonance
are unlikely.

= 33.7 ups or 2022 upm

Uneven Distribution of Pressure

Non-uniform pressure distribution around the circumference of the screw can lead to
vertical and horizontal forces of such magnitude, that the screw deflects into the barrel.
Even a pressure difference of 10 bar could create a horizontal force F, in an extruder
(diameter D = 150 mm within a section of length L = 150 mm)

R, =Ap-D-L=10-10°-0.15* = 225kN

According to RAUWENDAAL [8], the non uniform pressure distribution is the most probable
cause of screw deflection.

5.3 injection Molding

Other than extrusion, injection molding runs discontinuously and therefore the stages
involved in this process are time-dependent [14]. The quantitative description of the
important mold filling stage has been made possible by well known computer programs
such as MOLDFLOW [15] and CADMOULD [16]. The purpose of this section is to
present the basic formulas necessary for designing injection molding dies and screws on
a rheological and thermal basis and illustrate the use of these formulas with examples.
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5.3.1  Pressure Drop in Runner

As the following example shows, the pressure drop along the runner of an injection mold
can be calculated from the same relationships used for dimensioning extrusion dies.
Example

For the following conditions, the isothermal pressure drop Ap, and the adiabatic pressure
drop Ap are to be determined:

For polystyrene with the following viscosity constants according to Equation 1.36,
Section 1.3.7.3:

Ay = 44475

A, = —0.4983

A, = —0.1743

A; = 0.03594

A, = 000219

6 = 4285

¢ =133.2

Ty =190°C
flow rate nt = 330.4kg/h
melt density pn= 112 g/em®
specific heat com= 1.6 KJ/tkg - K)
melt temperature T =230°C

length of the runner L = 101.6 mm
radius of the runner R = 5.08 mm
Solution

a) Isothermal flow

¥, from Equation 1.19:

_4Q 4-330.0

=—== - =795.8 st
R 36-7-1.12-0.508

1A
(Q = volume flow rate cm®/s)

ar from Equation 1.35:

= {T-T5} —4,285(230-150)
aT = 10C2 +{T-Tg) = 10 133.2+4(230-1%0} = 10—0.9896 = 0‘102

n from Equation 1.37:
n=3.956

1, from Equation 1.36:
n,=132Pa-s
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7 from Equation 1.22:
7=105013.6 Pa

K from Equation 1.26:
K=9911-10"%

Die constant G, from Equation 5.3:

=1.678-107°

L -
;r]s.ase (5.08-107°)59%

G. .. ==
circle (4 2.0.1016

Ap, with Q = 8.194 107 m*/s from Equation 5.2:
1

5. . 107%)5956
Apy = 10 (8.19% 1077) — 42 bar

(9.911-1072)59% . 1,678 - 10~

b) Adiabatic flow
The relationship for the ratio Ap is [17]
0
Apy -1
where
=1+ 280 (5.75)
m ° cpm

Temperature rise from Equation 5.18:

AT = 10-pip- Cpm - 10-1:4122 e oMK
For polystyrene

B=0026K"

2= 2.34- 0,026 = 1.061
Finally, Ap from Equation 5.74;

Ap = Bp, Ing, _42-In1061 _ ..\

7 -1 0.061

In the adiabatic case, the pressure drop is smaller because the dissipated heat is retained
in the melt.
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Examples of calculating pressure drop in runners of different geometry are shown in the

following figures:

100.00

Pressure drop (bar}
g

1.00

€

R=3mm

==

Estmm

= R = 9 mm

R=12mm
R=15mm

LDPE
T=200°C

Flow rate (kg/h)

L =100 mm

Figure 5.35 Pressure drop vs. flow rate for a circular cross-section for LDPE
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Figure 5.36 Effect of melt viscosity on pressure drop
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Figure 5.38  Pressure drop for a noncircular channel with R, = 2.77 mm and n = 2,052

5.3.2 Mold Filling

As already mentioned, the mold filling process is treated extensively in commercial
simulation programs [15, 16] and recently by Bancer [18]. In the following sections the
more transparent method of STEVENsoN [19] is given with an example.

5.3.2.1 Injection Pressure and Clamp Force

To determine the size of an injection molding machine for the production of a given
part, knowledge of the clamp force exerted by the mold is important, because this force
should not exceed the clamp force of the machine.

Injection Pressure

The isothermal pressure drop for a disc-shaped cavity is given as [19]

Apy lOs(l"”R)[N‘@“lﬂ'nR-rz'sz| b (5.76)

The fill time 7 is defined as {19]

_V-a

"5y (5.77)

T

The Brinkman number is given by [19]

2 . (I+ng }
Br=—0 K | Q360 (5.78)
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Example with symbols and units

The material is ABS with ng = 0.2565, which is the reciprocal of the power law exponent
1. The constant K, which corresponds to the viscosity 17, in Equation 5.52is K = 3.05 - 10%,

Constant injection rate Q =160 cm’/s

Part volume V =160 cm’

Half thickness of the disc b =21mm

Radius of the disc r, =120 mm
Number of gates N =1

Inlet melt temperature Ty=518K

Mold temperature Tw=323K

Thermal conductivity of the melt A =0.174 W/(m-K)
Thermal diffusivity of the polymer a =7.72-10" cm?/s
Melt flow angle [19] 0 =360°

The isothermal pressure drop in the mold Ap, is to be determined.

Solution
Applying Equation 5.76 for Ap,

4 L2653
Ap, = 53.05 10 360 -160 - (1 + 2 0.26525) _ [ 12 ) _ 254 bar
10° (1 -02655)} 1-360-47-12-0.105 0.105
Dimensionless fill time 7 from Equation 5.77:
-4
I 160 -7.72 l(: — 0.07
160 - 0,105
Brinkman number from Equation 5.78:
0.105" - 3.05 - 10* 160 - 360 e
Br=—0p—— - . =0.771
10° - 0.174- 195 1-360-2x-0.105° -12

From the experimental results of STEVENSON {19], the following empirical relation was
developed to calculate the actual pressure drop in the mold

h{ﬁJ =0337+4.717—-0093Br-267-Br (5.79)
P

The actual pressure drop Ap is therefore from Equation 5.79:

Ap =1.574 - Ap, = 1.574 - 254 = 400 bar
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Clamp Force

The calculation of clamp force is similar to that of the injection pressure. The isothermal
clamp force is determined from [19]

E(rz)=10-m§(;_"

R}@l (5.80)
where F,(r;} = isothermal clamp force (N).
F,(r,) for the example above is with Equation 5.80

1-0.2655

E(r)=10-712°
(1) [3—0.2655

]' 254 = 308.64 kKN

The actual clamp force can be obtained from the following empirical relation, which was
developed from the results published in [19].

In{F/F)=0372+7.67—0.084 Br —3.538 ¢ Br (5.81)
Hence the actual clamp force F from Equation 5.81
F=191-308.64 = 589.5 kN

The above relationships are valid for disc-shaped cavities. Other geometries of the mold
cavity can be taken into account on this basis in the manner described by Stevenson
[19].

5.3.3  Flowability of Injection Molding Resins

The flowability of injection molding materials can be determined on the basis of melt
flow in a spiral channel. In practice, a spiral-shaped mold of rectangular crosssection
with the height and width in the order of a few millimeters is often used to classify the
resins according to their flowability. The length L of the solidified plastic in the spiral is
taken as a measure of the viscosity of the polymer concerned.

Figure 5.39 shows the experimentally determined flow length L as a function of the height
H of the spiral for polypropylene. A quantitative relation between L and the parameters
influencing L such as type of resin, melt temperature, mold temperature, and injection
pressure can be developed by using the dimensionless numbers as defined by THornE
[23] in the following manner:

The Reynolds number Re is given by [23]

_ VeZ-nR . p’ H*HR
kn

Re (5.82)

where
K=n (5.83)
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Figure 5.39 Fiow length L as a function of the spiral height H
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oy
. 6-Q
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H =05H
Prandtl number Pr [23]
K e, HUT
Pro-—f (5.84)
AV, TR
and Brinkman number Br [23]
* ylemg | prt-ng)
=k Ve -H (5.85)
In addition, the Graetz number is defined by
Go=Sfe (5.86)
z= .
A-L

As shown in [20] and in Figure 5.40, the Graetz number correlates well with the product
Re:Pr-Br
Gz=f(Re-Pr-Br) (5.87)

An explicit relationship for the spiral length L can therefore be computed from this
correlation.
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Figure 540 Dimenstonless groups for determining the flowability of a resin [20]

Symbols and units:

Br Brinkman number

% Specific heat kJ/(kg - K)

¥  Apparent shear rate s

G Mass flow rate kg/h

Gz Graetz number

H  Height of the spiral mm

H'  Half height of the spiral mm
k' Constant from Equation 5.83
L Length of the spiral mm

np  Reciprocal of the power law exponent
Pr  Prandtl number

Q  Volume flow rate m’/s

Re Reynolds number

Ty Melt temperature °C

Tw Mold temperature °C

Velocity of the melt front m/s

W Width of the spiral mm

A Thermal conductivity W/{m - K)
g Melt density g/cm®

n, Melt viscosity Pas

Example

This example illustrates the calculation of the dimensionless numbers Gz, Re, Pr and Br
for:

W=10mm; H=2 mm; L =420 mm; p=1.06 g/cm3; =2 kJ/(kg - K);
A=15W/(m - K); Ty = 270 °C; Tyy = 70 °C; G = 211.5 kg/h

Resin-dependent constants according to Equation 1.36:

Ag=4.7649; A, = 04743; A, = 0.2338; A, = 0.081; A, = 0.01063;
¢ =445 ¢,=146.3; T, =190 °C
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Solution

The conversion factors for the units used in the calculation of the dimensionless numbers
below are

F, = 0,001; F, = 1000; F;=3600
The Graetz number Gz is calculated from
E-G-¢
Gz=—2—P°
F-F-A-L

with G in kg/h and L in mm. Using the values given above, Gz =186.51. The Reynolds
number is obtained from

Reo B Ve - p H™
k!

with V, in m/s, H' in m and pin g/cm’.

Using the values given above, Re = 0.03791.
With H in m and V, in m/s we get from
E -k - ¢, - )
R
and the Brinkman number Br from

Pr =103302.87

K ovirre . gritom)
T ATy —Ty)

Finally, the product Re - Pr - Br = 7768.06.

Br = 1.9833

5.3.4 Cooling of Meltin Mold

As mentioned in Section 3.2.1, the numerical solution of the Fourier equation, Equation
3.31, is presented here for crystalline and amorphous polymers.

5.3.4.1 Crystalline Polymers

The enthalpy temperature diagram of a crystalline polymer shows that there is a sharp

enthalpy rise in the temperature region where the polymer begins to melt. This is caused
by the latent heat of fusion absorbed by the polymer when it is heated and has to be
taken into account when calculating cooling curves of crystalline polymers.

By defining an equivalent temperature for the latent heat (Figure 5.41), GLoor [22]
calculated the temperature of a slab using the Fourier equation for the non-steady-state
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Figure 5.41 Representation of temperature correction for latent heat [22]

N\

heat conduction. The numerical solution of Equation 3.31 using the correction introduced
by Groor [22] was given in [25] on the basis of the method of differences after ScHMIDT
[24]. A computer program for this solution is presented in [3]. The time interval used in

this method is

c. - 2
At = "AP-% (5.88)

where

At = time interval

Ax = thickness of a layer

M = number of layers, into which the slab is devided, beginning from the mid plane of
the slab (Figure 5.42)

The mold temperature and the thermodynamic properties of the polymer are assumed
to be constant during the cooling process. The temperature at which the latent heat is
evolved, and the temperature correction W, (Figure 5.41) are obtained from the enthalpy
diagram as suggested by Gioor [22]. An arbitrary difference of about 6 °C is assigned
between the temperature of latent heat release at the mid plane and the temperature at
the outer surface of the slab.

h h h h h ki
Figure 5.42 Nomendature for numerical solution of non-steady state conduction in a slab [25]
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Figure 5.43 shows a sample plot of temperature as a function of time for a crystalline

polymer.
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5.3.4.2 Amorphous Polymers

Amorphous polymers do not exhibit the sharp enthalpy change as crystalline plastics
when passing from liquid to solid. Consequently, when applying the numerical method
of ScHMIpT [24], the correction for the latent heat can be left out in the calculation,
A sample plot calculated with the computer program given in [3] is shown in Figure 5.44
for amorphous polymers.

It is to be mentioned here that the analytical solutions for non-steady heat conduction
given in Section 3.2.1 serve as good approximations for crystalline as well as for
amorphous polymers.

5.3.5 Design of Cooling Channels
5.3.5.1 Thermal Design

In practice, the temperature of the mold wall is not constant, because it is influenced by
the heat transfer between the melt and the cooling water. Therefore, the geometry of the
cooling channel lay out, the thermal conductivity of the mold material, and the velocity
of the cooling water affect the cooling time significantly.

The heat transferred from the melt to the cooling medium can be expressed as (Figure
5.45)

Quy =107 - [(Tyy = Te) Cpg + i) P - % - x (KJ/m) (5.89)
The heat received by the cooling water in the time # amounts to

1, _31
A S, a-10°-2-7-R

-1
Q, =107 -t ( ) (T = Toter) (KV/m)  (5.90)

The cooling time f in this equation can be obtained from Equation 3.41. The influence
of the cooling channel lay out on heat conduction can be taken into account by the shape

factor S, according to [23, 26].
/
D

] /d ey
N
///’/

X,
e i,

Figure 5.45 Geometry for the thermal design of cooling channels
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S = Py 4
. =
2-X-sinh(2n—y]
l.n X
wd

(5.91)

With the values for the properties of water

¢ =418 KJ/(kg- K)
U=112-10%Pa-s
A=0.6W/(m-K)

the heat transfer coefficient a can be obtained from Equation 3.52

0.031395
= Tg B Reo's

(5.92)

The mold temperature T, in Equation 3.41 is calculated iteratively from the heat balance

Qab = Qw .

Example with symbols and units

Part thickness

Distance

Distance

Diameter of cooling channel

Melt temperature

Demolding temperature

Latent heat of fusion of the polymer
Specific heat of the polymer

Melt density

Thermal diffusivity of the melt
Kinematic viscosity of cooling water
Velocity of cooling water
Temperature of cooling water
Thermal conductivity of mold steel

$ =2 mm
x =30mm
¥ =10mm
d =10mm
Ty =1250°C
T, =%0°C

in =130KkJ/kg

Cps = 25K/ kg - K)

pn =079 glem’
=83.107* em%s

a

v =12-10"°%m?s
¥ =1m/s

Tt =15°C

Ay, =45WHm - K)

With the data above the heat removed from the melt Q, according to Equation 5.89 is

2

Q,, =107 [(250 — 90) - 2.5+ 130] - 0.79 - > 30 = 12.56 k}/m
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Shape factor S, from Equation 5.91:

S = iz = 3.091

2.39sinh[2”'10)

In
10

Reynolds number of water:

-3
Re=10‘3-u-dfv=&l—f=8333
12-10

Using Equation 5.92 for the heat transfer coefficient o

_ 0.031395 - Re™®

= = 4300 W/(m* -K
vater 107210 ( )

From Equation 5.90 we get for the heat received by the cooling water Q,,

-1
- 1 1
Q. =107 ¢ + (T, —15)
. (45 +3.091 4300107 27 SJ
Cooling time # from Equation 3.41:
i (0.2 |2, @250-T,)
K™ 22.83.10% |z (90-T,)
From the heat balance
Qu = Q,, = 12.56
we obtain by iteration
T,=37.83°C
Finally, the cooling tite fy with T, = 37.83 is from Equation 3.41
te=8.03s

The influence of the cooling channel lay out on cooling time can be simulated on the
basis of the equations given by changing the distances x and y (Figure 5.45) as shown
in Figure 5.46 and Figure 5.47. The effects of the temperature of cooling water and of
its velocity are presented in Figure 5.48 and Figure 5.49, respectively. From these results
it follows that the cooling time is significantly determined by the cooling channel lay

out,
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5.3.5.2 Mechanical Design

The cooling channels should be as close to the surface of the mold as possible so that heat
can flow out of the melt in the shortest time possible.

Mold surface

T4 4~

ANANN

i

&,

Figure 5.50 Geometry for the mechanical design of cooling channels

However, the strength of the mold material sets a limit to the distance between the cooling
channel and the mold surface. Taking the strength of the mold material into account, the
allowable distance 4 (Figure 5.50) was calculated by [27] on the basis of the following
equations:

05-P-
bemax = dZ (5.93)
0.75-P-1
max —d (5.94)
1000-p-F P 0.15
= - + 5.95
where
P = mold pressure N/mm’
I,d = distances mm, see Figure 550
E = tensile modulus Nfmm®
G = shear modulus N/mm?

Oy, = allowable tensile stress N/ mm’

T, = allowable shear stress N/mm®

fmax = max. deflection of the mold material above the cooling channel um

The minimization of the distance d such that the conditions
f £ foon
-
T T

are satisfied, can be accomplished by the computer program given in [3]. The results of a
sample calculation are shown Table 5.2.
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Table 5.2 Results of Optimization of Cooling Channel Distance in Figure 5.50

Input Output

Mold pressure P = 4.9 Nfmm’ Channel distance 4 =2.492 mm
Maximum deflection  f =2.5pum Deflection f =2487pm
Modulus of elasticity  E = 70588 N/mm’ Tensile stress & = 39.44 N/mm’
Meodulus of shear G = 27147 N/mm’ Shear stress 7 =14.75 N/mm®
Allowable tensile stress &, = 421.56 N/mm'

Allowable shear stress rmm =294.1 N/mm’

Channel dimension ! =10 mm

The equations given provide approximate valaes for circular channels as well. The distance
from wall to wall of the channel should be approximately the channel length ! or channel
diameter, taking the strength of the mold material into account.

5.3.6 Melting in Injection Molding Screws

The plastication of solids in the reciprocating screw of an injection molding machine is
a batch process and consists of two phases. During the stationary phase of the screw
melting takes place mainly by heat conduction from the barrel. The melting during screw
rotation time of the molding cycle is similar to that in an extrusion screw but instationary.
With long periods of screw rotation, it approaches the steady state condition of extrusion
melting.

5.3.6.1 Melting by Heat Conduction

According to Donovan [28], the equation describing conduction melting can be written as

o 220 (T —T,) - A - exp(K? /4 G)  (5,=T) - 4, - exp(=K’/a)) (5.96)
i Do J7 o a, efiK2+/a) J7-a, erfc(Ki2a) '
where

T = temperature °C

A = thermal conductivity W/(m - K)
K = Parameter defined by [28] m/s>3
a = thermal diffusivity m’/s

I = latent heat of fusion kJ/kg

p = density g/cm’

Indices:

r: middle of solid bed
s: solid

m: melt

b: barrel
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The parameter K can be determined iteratively by means of the computer program given
in [9].

5.3.6.2 Melting during Screw Rotation

Analogous to the melting model of Tadmor (Section 4.2.3), DonovaN [28] defines an
arearatio A
A=t
Ar
to quantitatively describe the melting or solid bed profile of a reciprocating screw. A is

the ratio of the cross-sectional area of solid bed A, to the cross sectional area of screw
channel Ar.

The equations according te Donovan [28] for calculating the solid bed profiles are as

follows:
52
AflK tT_tR“‘(flJ_b‘i]
A — A = (5.97)
. Af — A f-2x-N-t )
= A - f - R .
~ l—exp{_ﬁ'z"r'N'tR) exP( 60 58)
L 60 |
S A -4
A = A, - (_ﬁ'm'N'tR] (5.99)
L 60 ]
where

t; = total cycle time s

tp = screw rotation time s

8, = thickness of melt film m

H = depth of the screw channel m
b = dimensionless parameter

N = screw speed rpm

Indices:

i: beginning of screw rotation
f.  end of screw rotation
e extrusion
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Figure 551  Solid bed profile of an injection molding screw
The thickness of the melt film & and the solid bed profile for steady state extrusion A,
can be obtained from the relationships in the Tadmor model given in Section 4.2.3.

The area ratio at the start of screw rotation A; and the value at the end of screw rotation
A; can then be obtained by using the computer program given in [9]. Figure 5.51 shows
the solid bed profiles of a computer simulation [9] for a particular resin at given operating

conditions.

Calculation procedure
Step 11 Calculate K using Equation 5.96.
Step 2: Calculate 8, according to

5 o5 B =T+ v} 107w
v . 10° Vbx Pm [cps (Tm_rs)"'im]

The average temperature in the melt film is obtained from

T. =05 (T, :r)+10_4ﬂ'fvjz
| b n IZAm
Substitute & with J,,,.

tep 3: Calculate the solid bed ratio A, for steady-state extrusion with the simplified
model for a linear temperature profile.

Step 4: Find the solid bed ratio Ay at the end of the screw rotation using Equation 5,97
and Equation 5.98.

Step 5. Calculate A, the solid bed ratio at the start of screw rotation from Equation 5.99.
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The following sample calculation shows the symbols and units of the variables occurring

in the equations above,

Example

The thermal properties for LDPE and the barrel femperature are as given in the previous

calculation for the parameter K. In addition,

Total cycle time tr =45s

Screw rotation time th =225
Empirical parameter for all polymers B =0.005
Screw speed N =56rpm
Channel depth H =9%8mm
Channel width W = 52.61 mm
Cross-channel velocity of the melt Vx = 5.65 cm/fs
Relative velocity of the melt v, =15.37 cm/s
Solids temperature T, =20°C

By using these values and by iteration, the following target values are obtained:

Melt viscosity in the film

f =211Pa-s

Average temperature of the melt in the film  T; = 172.8°C
Average thickness of the melt film 8, = 7678107 cm

Using K=5.6- 10"1 the solid bed ratios are found to be: the solid bed ratio at the end of
screw rotation, Ar = 0.583 ; the solid bed ratio at the start of screw rotation, Al =0.429.
The solid bed ratio for steady-state extrusion, A, , is calculated from the simplified melting
model for extrusion. Its numerical value for the conditions above is A =0.75.

In the following figures the steady-state extrusion profile begins at the position of the
stroke. The temperature of the melt refers to the temperature at the end of the screw
for the case of steady-state extrusion. The solid bed ratio A" is the ratio between the
cross-sectional area of the solid bed A, and the total cross-sectional area of the channel
Ar. In Figure 5.52 the effect of the resin type on the solid bed profiles is presented. It
appears that the conductivity parameter K and the melt viscosity affect these profiles
significantly, even if the screw rotation and cycle times remain the same. It can be seen
from Figure 5.53 that the barrel temperature has little effect on the plastication process
in the screw.

As Figure 5.54 depicts, slow screw speed and a high percentage of screw rotation time
compared to total cycle time favor melting strongly, which has also been found by Donovan
[28]. The marked influence of screw geometry on melting becomes clear from Figure 5.55.
As can be expected, melting is much faster in a shallower channel.
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Figure 5.52 Effect of polymer on the melting profiles
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Figure 5.54
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Figure 5.55 Effect of screw geometry on the melting for LDPE

5.3.7 Predicting Flow Length of Spiral Melt Flows

Injection molding is widely used to make articles out of plastics for various applications.
One of the criteria for the selection of the resin to make a given part is whether the resin
is an easy flowing type or whether it exhibits a significantly viscous behavior. To determine
the flowability of the polymer melt, the spiral test, which consists of injecting the melt
into a spiral shaped mold shown in Figure 5.56, is used. The length of the spiral serves as
a measure of the ease of flow of the melt in the mold and enables mold and part design
appropriate for a specific material flow behavior.

Cross section

Figure 5.56 Schematic representation of spiral form
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Increasing MFI

Flow length L

Spiral height H

Figure 5.57 Schematic flow curves

The experimental flow curves obtained at constant injection pressure under given melt
temperature, mold temperature, and axial screw speed are given schematically in
Figure 5.57 for a resin type at various spiral heights with melt flow index of the polymer
brand as parameter. By comparing the flow lengths with one another at any spiral height,
also called wall thickness, the flowability of the resin in question with reference to another
resin can be inferred.

The transient heat transfer and flow processes accompanying melt flow in an injection
mold can be analyzed by state-of-the-art commercial software packages. However, for
simple mold geometries, such as the one used in the spiral test, it is possible to predict
the melt flow behavior on the basis of dimensionless numbers and obtain formulas useful
in practice. These relationships can easily be calculated with a handheld calculator oftering
quick estimates of the target values. Owing to the nature of non-Newtonian flow, the
dimensionless numbers used to describe flow and heat transfer processes of Newtonian
fluids have to be modified for polymer melts. As already presented in Section 5.3.3, the
movement of a melt front in a rectangular cavity can be correlated by Graetz number,
Reynolds number, Prandtl number, and Brinkman number. Because the flow length in a
spiral test depends significantly on the injection pressure (Figure 5.58), the Euler number
[41] is included in the present work in order to take the effect of injection pressure on
the flow length.
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Figure 5.58 Effect of injection pressure on flow length

In addition to the dimensionless numbers Gz, Re, Pr, and Br, we consider [41] the Euler
number

Eu= 0021 (5.100)
£ Ve
where p is the injection pressure.

Experimental flow curves for four different resins measured at constant injection pressure
under different processing conditions and spiral wall thicknesses are given in Figure 5.59.

5 LDPE s HDPE
E 20 LT E 0 T
é - T i "E, - "
= 30 ] _!_ 3 ay =
g | g
600 900 1200 1500 500 900 1200 1500
Injection pressure {bar} Injection pressure {(bar)
PP PS
- 50 - £ 50
£ L ~ E "
? —_ ? : o
2 g9 2 30
3 g
g 20 : [TH 20 L
800 960 1200 1500 600 900 1200 1500
Injection pressure {bar) Injection prassure (bar)

Figure 5.59 Experimental flow curves for LDPE, HDPE, PP, and P$
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Figure 5.60 Graetz number as a function of the product of Re, Pr, Br, and Eu

The flow length as a function of injection pressure is shown in Figure 5.58 for LDPE as
an example. The Graetz numbers calculated from the experimentally determined spiral
lengths at different operating conditions and resins are plotted as functions of the product
Re - Pr - Br - Eu as shown in Figure 5.60. As can be seen from this figure, the correlation
of the Graetz number with this product is good and thus for any particular material the
spiral length can be predicted from the relationship

Gz = f{(Re - Pr- Br - Eu) (5.101)

Figure 5.61 shows the good agreement between measured and calculated spiral lengths
for the experimentally investigated resins.
Sample Calculation

The example given below shows how the flow length of a given resin can be calculated from
Equation 5.101: W= 10 mm, H =2 mm, p = 1.06 g/cma,cp =2k)/kg-K,A=15W/K-m,
Ty = 270°C, Ty = 70°C and G = 211.5 kg/h.

The melt viscosity 1, is calculated from

M =lgap + Ay + Ay - lglag - P) + A, -[lglag - T
+ Ay -[lglar - T + A, -liglay - DI
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Figure 5.61 Comparison between measured and calculated flow length {(— calculated; D measurad)

The shear rate § is obtained from

___6-Q
W - H?

mean
where W =W+ H - tan

Ay A, Ay Ay, A, are material constants, and ag the shift factor. For amorphous polymers
the shift factor is obtained from

_—0 (T-T)

Iga
B S L, A (T-T))

with constants ¢, ¢; and melt and reference temperatures T and T, respectively, in K. For
semi-crystalline and crystalline polymers ar is calculated from

. =bl-eb2”-

with the constants b,, b, and melt temperature Tin K.

Using the values of A, = 4.7649, A, = —0.4743, A, = —0.2338, A, = 0.081, 4, = —0.01063,
6, =4.45, ¢, = 146.3 and T = 190 °C, the following output is obtained: Re = 0.05964,
Pr = 76625,34, Br = 1.7419, and Eu = 10825.84. The Graetz number Gz for the product
Re - Pr - Br - Eu follows from Figure 5.60: Gz = 217.63. Hence [ = 420 mm.
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Nonlinear Viscoelastic
Behavior
tensile compliance
tensile extensional flow
tensile relaxation
modulus
transient state

Nusselt number
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Part failure
Peclet number
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Plastics parts
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Radiation

Recoverable shear strain
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Screw extruders
mechanical design
scale-up

Shear
compliance
time-dependent behavior

Shear flow
steady

Shear rate
apparent

Shear Stress

Sherwood number

Shift factor

Solids
conveying
conveying efficiency

Specific heat

Spiral melt flows

Stefan-Boltzmann
constant

Stokes number

T

Temperature fluctuation
Tensile creep compliance
Tensile stress

Thermal conductivity
Torsion

Trouton viscosity
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\4 Muenstedt
Viscosity power law EI

apparent Viscosity, influence of

true mixture
Viscosity function molecular weight

Carreau pressure

Klein shear rate
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